ECONSTOR Make Your Publications Visible.

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Masuku, Khumbulani L.; Gopane, Thabo J.

Article

Technical trading rules' profitability and dynamic risk premiums of cryptocurrency exchange rates

Journal of Capital Markets Studies (JCMS)

Provided in Cooperation with: Turkish Capital Markets Association

Suggested Citation: Masuku, Khumbulani L.; Gopane, Thabo J. (2022) : Technical trading rules' profitability and dynamic risk premiums of cryptocurrency exchange rates, Journal of Capital Markets Studies (JCMS), ISSN 2514-4774, Emerald, Bingley, Vol. 6, Iss. 1, pp. 6-32, https://doi.org/10.1108/JCMS-10-2021-0030

This Version is available at: https://hdl.handle.net/10419/313289

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.



https://creativecommons.org/licenses/by/4.0/

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU

The current issue and full text archive of this journal is available on Emerald Insight at: https://www.emerald.com/insight/2514-4774.htm

JCMS 6,1

6

Received 7 October 2021 Revised 1 December 2021 14 December 2021 Accepted 8 January 2022

Technical trading rules' profitability and dynamic risk premiums of cryptocurrency exchange rates

Khumbulani L. Masuku and Thabo J. Gopane Department of Finance and Investment Management, University of Johannesburg, Johannesburg, South Africa

Abstract

Purpose – The study considers time-varying risk premium in investigating the capability of technical analysis (TA) to predict and outperform a buy–hold strategy in Bitcoin exchange rate returns.

Design/methodology/approach – The study tests the technical trading rule of fixed moving average (FMA) on daily actual and equilibrium returns of Bitcoin exchange rates. The equilibrium returns are computed using dynamic CAPM in conjunction with a VAR-MGARCH (1, 1) system. The empirical evaluation of the study uses a case study of four Bitcoin exchange rates (BTC/AUD, BTC/EUR, BTC/JPY and BTC/ZAR) for the period 19 June 2010 to 30 October 2020.

Findings – The findings are consistent with related studies in conventional foreign exchange markets that find TA to be profitable, especially in emerging markets. Nevertheless, the consideration of risk premium has the effect of reducing the abnormal returns. Also, further robust tests reveal that Bitcoin returns possess a momentum effect which prompts further study in efficient market hypothesis research.

Practical implications – The empirical findings of this study should benefit portfolio managers and active investors on the strength of TA to predict returns in a speculative market like the Bitcoin exchange rate market. **Originality/value** – The study takes cognisance that cryptocurrency trading is speculative in nature which renders it a good candidate for TA methods. While there are studies that have explored the value of TA in Bitcoin exchange rates, these studies fail to incorporate the effects of time-varying risk premiums, the strength and focus of the current paper.

Keywords Bitcoin, Cryptocurrency, Efficient market hypothesis, Fixed moving averages, Profitability, Technical analysis

Paper type Research paper

1. Introduction

Over the years, academics and practitioners have demonstrated an overwhelming interest in the profitability of technical analysis (TA) on practically all financial systems and assets. Historical evidence indicates that technical trading strategies were profitable in foreign exchange markets and future markets (Smidt, 1965; Sweeney, 1986; Taylor, 1986), but not in stock markets (Fama and Blume, 1966; Van Horne and Parker, 1967). More recent empirical studies suggest that technical trading rules (TTRs) may generate positive profits in certain speculative markets, most notably in foreign exchange and futures markets (Nazário *et al.*, 2017). Various theoretical and empirical explanations have been proposed for TA-based profits. From the investment theory perspective, Neely *et al.* (2014) suggest that abnormal



Journal of Capital Markets Studies Vol. 6 No. 1, 2022 pp. 6-32 Emerald Publishing Limited 2514-4774 DOI 10.1108/JCMS-10-2021-0030

[©] Khumbulani L. Masuku and Thabo J. Gopane. Published in *Journal of Capital Markets Studies*. Published by Emerald Publishing Limited. This article is published under the Creative Commons Attribution (CC BY 4.0) license. Anyone may reproduce, distribute, translate and create derivative works of this article (for both commercial and non-commercial purposes), subject to full attribution to the original publication and authors. The full terms of this license may be seen at http://creativecommons.org/licences/by/4.0/legalcode

profits based on TA trading may arise because of market "frictions", such as noise in current equilibrium prices, traders' sentiments, herding behaviour, market power or chaos. Other possible explanations for the persistence of trading profits are the presence of central bank interventions (particularly in foreign exchange markets), order flow, temporary market inefficiencies, risk premiums, market microstructure deficiencies or data snooping (Park and Irwin, 2007). Regardless of the origin or description of profits, the current study only asks the question of whether TA trading is profitable, particularly in the Bitcoin case.

Cryptocurrencies may show connectedness to currency markets more than stock, and this means they may mimic some of the characteristics that make the foreign exchange market attractive to TA users. Extant literature shows that researchers have investigated the profitability of TTRs in a variety of markets for the purpose of either uncovering profitable trading rules or testing market efficiency, among other things. However, sufficient in-depth analysis of the cryptocurrency market is still lacking, so that the stylised facts of related TA benefits are yet to be understood. The current study contributes towards closing this knowledge gap. The objective of this paper is to investigate the profitability of TA on four Bitcoin exchange rates (BTC/AUD, BTC/EUR, BTC/IPY and BTC/ZAR) after accounting for time-varying risk premiums. Furthermore, the study evaluates the presence of a momentum effect in Bitcoin exchange rates. While most papers focus on Bitcoin as an individual currency (Urguhart, 2016; Bariviera, 2017; Khuntia and Pattanavak, 2018; Kristoufek and Vosvrda, 2018; Tiwari et al., 2018; Sensoy, 2019), the current study investigates the TA profitability in four Bitcoin exchange rates for comparison and breadth. The current study extends and deepens the research on TA usage in foreign exchange markets (Kho, 1996) to cryptocurrency exchange rate. This study will test TA profitability in Bitcoin by employing 24 Fixed Moving Average (FMA) techniques.

The observed results were compared to a buy-hold strategy accordingly, and a significant difference was noted. The study's findings indicate that the chosen TTRs' strategies outperform the buy-hold strategy. A portion of the reported gains can be accounted for by time-varying risk premiums, as profits decline when risk is included. In view of the current and related studies, it is evident that Bitcoin may be an ideal laboratory for experimenting with TA since it lacks discernible fundamentals. The study advanced a novel stylised fact, indicating that the MA trading rule can forecast Bitcoin returns even when time-varying risk premiums are taken into consideration.

The remainder of this paper is structured as follows: Section 2 provides a literature review, Section 3 introduces the historical background of Bitcoin as a cryptocurrency, Section 4 describes Bitcoin's possible candidature of TA, Section 5 outlines the data and methodology utilised in the study, Section 6 presents the empirical results, Section 7 presents a discussion of results and Section 8 concludes the study.

2. Literature review

2.1 Related studies

According to the efficient market hypothesis (EMH), stock prices fully contain all relevant market information. This means that any new information should be reflected in the security prices as soon as it becomes available. To classify the type of information captured by stock prices, there are three classes of market efficiency: (1) strong form like private firm information, (2) semi-strong form such as published economic statistics and financial statements, as well as (3) weak form which is primarily historical security prices and trading volume. Gerritsen *et al.* (2021) discovered that Bitcoin price behaviour has intertemporal predictability, implying that investors may estimate future gains. Trading rules based on past data should not be successful according to weak-form market efficiency. TTRs such as filter rules (Alexander, 1961, 1964; Fama and Blume, 1966; Sweeney, 1988), relative strength rules (Levy, 1967; Jensen and Benington, 1970; Brush and Boles, 1983; Jacobs and Levy, 1988)

Technical trading rules' profitability

and moving average trading rules (Van Horne and Parker, 1967; Dale and Workman, 1980) have been shown in earlier empirical studies to be unable to perform better than the traditional buy–hold strategy, and that any predictive variation in security returns is statistically and economically insignificant.

Recent research, on the other hand, has shown that simple trading strategies can be useful in predicting stock market returns. For example, an influential TA study by Brock *et al.* (1992) investigated two simple TTRs, namely, Moving Average (MA) rules as well as Trading Range Breakout (TRB) rules, and found that these two simple rules had considerable predictive power for the US equity index returns. These results were extended and confirmed by Bessembinder and Chan (1998) using US data, and Bessembinder and Chan (1995) using data from Asia–Pacific stock markets. The studies made an important observation that TA has less explanatory power in the more developed markets, but is more successful in emerging markets (Ito, 1999; Ratner and Leal, 1999; Miller *et al.*, 2019; Grobys *et al.*, 2020). Therefore, TA profitability (or its possibility) may be seen as one of the EMH anomalies arising from weak form efficiency.

Regarding the cryptocurrency market, the focus area of the current paper, there are several studies that have explored the implications of TA or its relevance in investment analysis. Miller *et al.* (2019) examined an identification technique in automated price patterns for Bitcoin cryptocurrency based on 1-min price data. The research examined several well-known TA patterns and developed trading procedures for and evaluated the effectiveness of selected trading strategies. The findings indicate that employing smoothing splines to deduce TA patterns has several advantageous and promising methodological properties. Additionally, the study discovers that methods based on certain TA patterns outperform the traditional buy-and-hold approach. A related study by Grobys *et al.* (2020) examined simple moving average trading strategies utilising daily price data from 2016 to 2018 for the 11 most-traded cryptocurrencies. Their findings indicated that the variable moving average technique is effective. When the average market return is considered, the TTR remains beneficial.

In a parallel study, Resta *et al.* (2020) used trend-following and mean-reverting techniques and found supportive results for the TA method when investigating the profitability of TTRs in the Bitcoin market. Using daily and five-minute interval data, their study period covered 1 January 2012 to 20 August 2019. Generally, the study concluded that daily trading data is more successful than trading intraday data. More meaningfully, the study revealed that simple Moving Averages techniques are superior when dealing with daily data. This study supports Gerritsen *et al.* (2020), who find TTRs to be more cost-effective than the buy–hold strategy when dealing with daily Bitcoin data. Other studies that reach favourable conclusions regarding the benefit of TA in cryptocurrency exchange rates include Tiwari *et al.* (2018), Miller *et al.* (2019), Gerritsen *et al.* (2020), and Detzel *et al.* (2021).

Given the increasing interest in cryptocurrency markets, including the spread into central bank digital currency, it is evident that the cryptocurrency market is enduring. The current study incorporates developments in the literature and expands knowledge on Bitcoin and TA profitability using more recent data, while applying the wisdom of tested techniques suggested by Brock *et al.* (1992) and Kho (1996).

2.2 Hypotheses

The current study examines three hypotheses to guide empirical investigation for TA profitability on four Bitcoin exchange rates (BTC/AUD, BTC/EUR, BTC/JPY and BTC/ZAR). The first hypothesis (hereafter coded, H1) states that when the mean returns derived from *TTR* are tested respectively on both equilibrium and actual returns, they are not different from the mean returns of normal trading days (buy–hold strategy). Simply put, this means that the buy (sell) returns of TTRs should not be different from the returns of normal trading days when there are no trading signals. The second hypothesis (H2) states that there is no

JCMS

difference in means between the buy and sell spread. This implies that if TTRs are not profitable, then the study should find no significant difference between the buy-and-sell returns. The next hypothesis is based on the momentum effect. Momentum is the rate of change of security prices or returns. If the rate of change of returns is high, momentum is regarded to be strong; if it is low, momentum is believed to be low (Jegadeesh and Titman, 1993; Gharaibeh *et al.*, 2021), and these patterned changes are exploitable for the predictability of security returns (Chan *et al.*, 2000). Therefore, the third hypothesis (H3) states that there is no relationship between price movements and return fluctuation (momentum effect). That is, the test enquires if further increases or decreases are associated with respective buy and sell signals.

3. Bitcoin's operational design

Bitcoin was birthed out of the concept of cryptocurrency which dates back to the 1980s when David Chaum wrote a seminal chapter on blind signature cryptographic primitives (Chaum, 1983). The author then proposed a useful cryptographic approach for application in safe digital currency transmissions. Cryptography, as defined by Harwick (2016), is a mechanism for creating virtual "money" and ensuring its secure ownership and transaction through the application of a cryptographic problem. Since its inception in 2009, Bitcoin as the first cryptocurrency has experienced both volatile and calm fluctuation over time.

Bitcoin is a decentralised digital currency system that is operated with blockchain technology. The production of Bitcoin is associated with a "mining" process (an equivalent of minting in fiat money), in which miners (or peers) use computers to create new Bitcoins by solving complex mathematical problems known as a *proof of work* or consensus method. Miners are rewarded with portions of Bitcoin for each Bitcoin created (Cheah and Fry, 2015). The Bitcoin structure is intended to remain unchanged from inception until the last Bitcoin is mined. The Bitcoin cash system is designed with a total capacity of 21 million (Shakya *et al.*, 2021). At the time of writing, there are 18,361,438 Bitcoins in circulation, leaving just 2,638,562 Bitcoins to be mined over the next approximately 119 years (Gandal and Halaburda, 2016). In view of the well-defined capacity and monetary system of Bitcoin, it is reasonable to conclude from the explained rules and empirical evidence (Gopane, 2019) that the worth of Bitcoin is remote from fundamental valuation. This creates a space for fundamental-free or speculative methods like TA to find direct opportunities for experimentation.

4. Bitcoin's speculative nature and technical analysis

The recurring characteristic of speculative assets, like Bitcoin, throughout history, has been due to their inability to be valued and/or unpredictable bubbles. Tulipmania, the South Sea Bubble, and others all signal speculation on one hand, and the difficulties of assigning an objective value to a speculative asset on the other (Garber, 1990). All speculative behaviours have been reflected in the exponential growth of the Bitcoin time series (Bariyiera *et al.*, 2017). Bitcoin lacks savings accounts and, as a result, no lending interest rates. The implication is that economic knock-on effects and spillover shocks are severely constrained or inapplicable. Consequently, the trails of fundamental analysis valuation are disabled significantly. Empirical assessment by Ciaian et al. (2016) finds that there are no macroeconomic indicators influencing Bitcoin's price, but they do not rule out the possibility that investor speculation has a major impact on the price development. The attraction of Bitcoin as an investment opportunity is significant in financial markets (Dyhrberg, 2016; Bouri et al., 2017) and this raises the need for appropriate price valuation methods that are empirically tested. In the instance of fundamental analysis irrelevance for Bitcoin, TA tools become a viable possibility. Therefore, the current study examines the profitability of technical trading methods and whether they are feasible alternatives as an analytical tool to evaluate Bitcoin exchange rate returns.

Technical trading rules' profitability

JCMS 5. Data and methodology

6.1

10

The research method in this study is quantitative and utilises several econometric equations, but the primary models are dynamic CAPM and VAR-GARCH. The modelling approach is explained later. First, we describe data collection and pre-validation procedures. The secondary dataset was sourced from the IRESS (2017) database and Bitcoincharts (n.d.). The frequency of the time series is daily and covers the period between 19 July 2010 and 31 October 2019. For empirical modelling purposes, the price data is converted into logarithmic returns through Eqn (1):

$$R_t = \log(P_t) - \log(P_{t-1}), \tag{1}$$

where R_t represents daily Bitcoin exchange rates at time t, while P_t is the price of Bitcoin. This formula is applied sequentially on each of the Bitcoin (BTC) exchange rates of Australia (BTC/AUD), Europe (BTC/EUR), Japan (BTC/JPY) and South Africa (BTC/ZAR). Table 1 presents the descriptive summary statistics for each of these exchange rates along with the global stock market benchmark, namely, the Morgan Stanley Capital International (MSCI) index. The MSCI is widely used as benchmark in academic research and practice (Hsu *et al.*, 2010; Bena *et al.*, 2017; Sermpinis *et al.*, 2021). The composition of the MSCI includes securities from 29 nations, and it is rebalanced on a quarterly basis. Therefore, the MSCI is deemed a suitable benchmark in the current study.

Table 1 summarises the time series used in the study and preliminary data validation. Judged by the positive means and stable standard deviations, the variables are comparable which is a useful feature in econometric analysis. The observation on the Jarque–Bera test, kurtosis and skewness reveals a typical phenomenon of financial time series (Brooks, 2014) of being skewed (positively or negatively) and peaked around the mean (leptokurtic). Other prevalidation tests performed on the data include Augmented Dickey–Fuller (ADF), which confirms that all variables are stationary which is desirable in regression analysis.

5.1 The design of technical analysis trading rules

The study employs 24 FMAs. The Moving Averages (MAs) were chosen because they are the most popular TTRs. Two types of MAs are used to produce buy-and-sell signals: a short-period average and a long-period average. In its basic design, this system is described as a buy (b) when the short-term moving average rises above the long-term moving average, and a sell (s) when the short-term moving average falls below the long-term moving average. Days when there is no signal are defined as buy–hold or neutral days (n). While varieties of MA rules are utilised in the literature, four popular specifications are 1/20, 1/50, 1/150 and

| | | BTC/AUD | BTC/EUR | BTC/JPY | BTC/ZAR | MSCI (world stock index) |
|---|---|--|--|---|---|--|
| | Mean Median Maximum Minimum Std. Dev Skewness | 0.005 0.003 0.361 -0.535 0.057 -0.206 | 0.005 0.003 0.354 -0.529 0.057 -0.193 | $\begin{array}{r} 0.005\\ 0.003\\ 0.361\\ -0.522\\ 0.057\\ -0.186\end{array}$ | 0.005 0.004 0.282 -0.274 0.051 0.098 | 0.000 0.001 0.027 -0.029 0.007 -0.267 |
| Table 1. Descriptive summary statistics | Kurtosis Jarque–Bera Probability Observations Source(s): Own | 14.172 12,624 0.000 2423 n computation | 14.221 12,731 0.000 2423 | 14.067 12,384 0.000 2423 | 8.196 2730 0.000 2423 | 4.716 326 0.000 2423 |

1/200. The complete list of MA rules used in this study is found in Table A1 (in Appendix) with and without a 1% band. The 1% band is used to reduce the number of false signals and is introduced around the short-term moving average. Urquhart *et al.* (2015) define buy-and-sell signals as:

$$\left[\sum_{\lambda=1}^{s} R_{t-(\lambda-1)}^{S}\right] > \left[\sum_{\lambda=1}^{L} R_{t-(\lambda-1)}^{L}\right] + \delta = \text{buy} \quad \delta \in \{0, 0.01,\},$$

where R_t is the return at time t, while L and S are the number of days for the long- and short-term moving average, respectively. The symbol, λ , represents the daily time index. Sell signals are generated when the inequality sign is reversed.

$$\left[\sum_{\lambda=1}^{s} R_{t-(\lambda-1)}^{S}\right] < \left[\sum_{\lambda=1}^{L} R_{t-(\lambda-1)}^{L}\right] + \delta = \text{sell}$$
(3)

The study, however, uses FMAs and holds signals for ten days before closing and re-entering the market. The ten-day holding period separates the FMA from the variable moving average (VMA), where positions are taken every day as signals are generated. These trading rules are in line with usage in Kho (1996) and Brock *et al.* (1992). The procedure helps minimise the possibility of statistical bias. The TTRs based on FMA are modelled using econometric models to assess the profitability of TA.

5.2 Econometric models

The study applies TA rules to two sets of Bitcoin exchange rate returns, namely, actual returns and equilibrium returns. The latter are computed using dynamic CAPM. In this modelling, and to make beta dynamic, the study employs the multivariate generalised autoregressive conditionally heteroscedastic (MGARCH) system.

5.2.1 Modelling equilibrium returns using dynamic CAPM. In cryptocurrency and capital market studies (Bouri *et al.*, 2020; Nugroho, 2021) there is a continual realisation that dynamic asset pricing modelling have a greater benefit over static models. The constant risk premium, or the static CAPM such as used by Sweeney (1986), is not deemed a suitable benchmark model for testing the trading rules' profitability because the risk of TA on the original series cannot essentially be comparable to that of the equilibrium strategy, among other things. Therefore, dynamic CAPM is the preferred model specified in Eqn (4):

$$R_{it} - R_{ft} = \alpha + \beta_{i,t} (R_{mt} - R_{ft}) + u_{it}, \quad u_t \sim N(0, \sigma^2)$$
(4)

$$\beta_{i,t} = \frac{\sigma_{im,t}}{\sigma_{m,t}^2} \tag{5}$$

In Eqn (4), the subscript *i* indexes the four Bitcoin exchange rates (BTC/AUD, BTC/EUR, BTC/JPY and BTC/ZAR). The disturbance term (u_t) is assumed to be normally distributed. The variables, R_{it} , R_{ft} and R_{mt} , are the returns of a financial asset, risk-free rate and the MSCI global market benchmark, respectively. The risk-free rate is proxied with the US's 90 days treasury bill (TBill) sourced from Federal Reserve Bank of St. Louis (Fred Database, n.d). The parameter, α , is the intercept evaluated in the model, while $\beta_{i,t}$ is the time-varying beta estimated in Eqn (5), where $\sigma_{im,t}$ is the covariance between market returns and exchange returns and $\sigma_{m,t}^2$ is the market return's variance. These two moments (variance and covariance) are computed through the VAR-MGARCH in Eqn (6). The VAR-MGARCH system is estimated through the mean Eqn (6a) of vector-autoregressive, VAR(p), and the variance equation by Baba *et al.* (1989), BEKK-MGARCH (p, q) in Eqn (6b):

2

11

Technical

trading rules'

(2)

profitability

$$y_t = \Pi y_{t-1} + \varepsilon_t \tag{6a}$$

$$H_t = CC' + A\varepsilon_{t-1}\varepsilon'_{t-1}A' + BH_{t-1}B'$$
(6b)

The parameters and standardised residuals are estimated using the maximum likelihood method. In the VAR(p) system (Eqn (6a)), y_t is a $k \times 1$ vector of exchange returns from Bitcoin exchange rates (BTC/AUD, BTC/EUR, BTC/JPY, BTC/ZAR), while Π is a $k \times k$ matrix of parameters to be estimated. In Eqn (6b), C is a $k \times k$ lower triangular matrix, while A and B are $k \times k$ coefficient matrices to be estimated. The disturbance term is assumed to be $\varepsilon_t \sim N(0, \Sigma_t)$, where Σ_t is the covariance matrix. The lag length for the equation is. p = q = 1.

To recapitulate, in tandem with Kho (1996) the dynamic CAPM (in Eqn (4)) is a mathematical formulation of log returns (Eqn (1)), and conditional beta (Eqn (5)) which is based on VAR-MGARCH system (Eqs (6a) and (6b)). Eqn (5) is a well-known relation of beta (see Brooks, 2014) and effectively captures covariation of Bitcoin exchange rates with the MSCI benchmark (introduced earlier). There are a few reasons why MSCI is deemed a viable proxy for global financial markets including cryptocurrency. First, virtual currencies are global in nature so the benchmark should have international positioning. Second, in the absence of a proved global index for cryptocurrency the literature (Neely *et al.*, 1997) shows that MSCI is a reasonable benchmark for foreign exchange market and by extension a close approximation or relevance for Bitcoin exchange rates in the current study. Other advantages of MSCI as a global benchmark is that it is widely applied in practice (Hsu *et al.*, 2010; Bena *et al.*, 2017) and academic research (Neely *et al.*, 2009; Bouri *et al.*, 2020), thus giving confidence for empirical usage. Another important feature of MSCI index is that it includes both large and small stocks thus mitigating the considerations of tradability and liquidity (Sermpinis *et al.*, 2021).

5.2.2 TA profitability: BLL's mean difference test. The Brock *et al.* (1992) (abbreviated, BBL) methodological approach is employed to study TA's individual (buy, sell and buy–hold strategy) mean returns. Brock *et al.* (1992) utilise the mean returns of trading rules and their respective *t*-statistic. The idea is to test the profitability of TA by comparing the difference between the conditional and unconditional mean returns from TA. The same technique is employed but with a robust approach instead of using the return on the buy–hold strategy. The return on days when no trading signal is emitted is represented by (n), which will be used to proxy the buy–hold days and when either a buy (b) or sell (s) signal is emitted. This test helps answer hypothesis H1. Brock *et al.* (1992) examined the predictive ability of MA using the student *t*-statistic ratio test to inspect whether the mean returns generated by TA are zero. The *t*-statistic of the differences between the means of *daily* and *buy–hold* returns is denoted by:

$$t_x = \frac{\mu_x - u}{\sqrt{\sigma^2 \left(\frac{1}{N} + \frac{1}{N_x}\right)}} \text{ where } x \in \{\text{buy, sell}\}$$
(7)

 μ_x and N_x represent the mean return and number of signals. σ^2 is the estimated variance. The parameter, u, is the mean return for buy–hold days.

5.2.3 TA profitability: Kho's mean-spread test. The objective of Eqn (8) is to test hypothesis H2, whether the spread between buy and sell signals is equal to zero. This special regression equation was first utilised by Cumby and Modest (1987) in a market-timing study. This test equation was first applied by Kho (1996) in TA profitability investigation. For this reason, we

JCMS 6.1 label Eqn (8) as *Kho's mean-spread test* for identification and simplicity. This equation is similar to Eqn (7), but it has advantages of validating the regression analysis with robust standard errors (Newey and West, 1987). Empirically, if the spread equals zero, it will mean that the buy and sell signals are equal. This would be a case against TA profitability. Based on the framework of Eqn (8), then if the spread is zero, the coefficient, α_1 will be insignificant.

$$R_{it} = \alpha_0 + \alpha_1 X_{it-1} + \varepsilon_{it} \tag{8}$$

In Eqn (8), R_t is the Bitcoin exchange rate return and X_{t-1} is the trading signal observed at day t-1. The parameters, α_0 and α_1 , are unknown coefficients to be estimated, while ϵ is the disturbance term which is assumed to follow normal distribution. The subscript, i, signifies Eqn (8) will be applied separately for each of the four Bitcoin exchange rate returns (BTC/AUD, BTC/EUR, BTC/JPY and BTC/ZAR). The regression equation is estimated separately for each type of signal. A signal is effectively a dummy, taking the value 1 if signal occurs and zero otherwise. Therefore, for buy signal, sell signal or both, the regressor, X_{t-1} takes the values X_{t-1}^b , $-X_{t-1}^s$ or $\{X_{t-1}^b - X_{t-1}^s\}$, respectively. By way of interpretation, a positive α_1 implies "... an average percentage increase in [daily] returns due to correct trading signals" (Kho, 1996, p. 258). Consequently, testing for the mean spread of buy–sell signals is equivalent to evaluating the null hypothesis, $\alpha_1 = 0$.

5.2.4 Time-varying risk premium and actual returns. The modelling objective of unconditional and conditional CAPM is to compare the equilibrium returns from both models based on the outcomes of buy and sell signals (TTRs). "If markets are efficient and the assumed model of asset pricing [CAPM] is correct, the technical rule returns in excess of the time-varying expected returns should have a zero conditional mean" (Kho, 1996, p. 279). This means that there should be no difference between expected returns under the two models. However, if the unconditional model outperforms the conditional model (or equal performance) it can be concluded that the abnormal profits derived from *TTR* are only reflecting time-varying risk premium. That is, investors who find TA profitable are merely compensated for their risk-bearing abilities. One way to determine the statistical significance of whether time-varying risk premium influences TTRs is to apply the standard *t*-test on the difference between the unconditional and conditional returns from *TTR* and examine if the spread is different from zero (Kho, 1996).

5.2.5 Consistency and robustness check. A robustness check in this study will examine the link between daily returns on Bitcoin and signals generated by the trading rules used (hypothesis H3). TA assumes that investors follow signals emitted at time t - 1, and that trades are executed at time t. That is, the demand (supply) of any particular security will exceed supply (demand) at time t, which will cause the price to either increase, thereby activating buy signals, or decrease to trigger sell signals at time t. The momentum effect is backed by one of the basic beliefs in TA, where the market is said to discount all types of information like fundamentals, non-fundamentals and even rumours (Masry, 2018). Since the early 1990s, the only other paper that has attempted to study the relationship between returns and signals is that of Bessembinder and Chan (1995) which applied a similar framework as in Eqn (9):

$$R_{it} = \beta_0 + \beta_1 FMA_{1,i,0} + \beta_2 FMA_{1,i,1} + \beta_3 FMA_{2,i,0} + \beta_3 FMA_{2,i,1} + \beta_4 FMA_{5,i,0} \beta_5 FMA_{5,i,1} + e_{it}$$
(9)

where $i \in \{20MA, 50MA, 150MA, 200MA\}$

In Eqn (9), β_0 is the regression intercept, R_t is the daily Bitcoin exchange rate return at day t, while FMA_i is the long-term trading signal emitted by a particular FMA. The hypothesis tests the link between signals released by MA rules and price fluctuations in the Bitcoin environment using the Ordinary Least Squares (OLS) model. In view of the above,

Technical trading rules' profitability the study hypothesis [3] is rejected if Eqn (9) is upheld, based on coefficient significance and F-statistical test.

6. Empirical results

This section presents the results of the paper. That is, an evaluation of the study hypotheses, H1, H2 and H3, relating to the mean-difference test (Eqn (7)), spread test (Eqn (8)) and momentum test (Eqn (9)), respectively. In addition, the analysis of risk premium and its implication in the results are elaborated. Before explaining the study results, it is important to provide a comprehensive report on post-estimation validation of the models in light of relevant econometric theories.

The practical timeseries regressions are generally affected by heteroscedasticity and autocorrelation problems regarding the BBL Model (Eqn (8)) and Momentum Effect test (Eqn (9)). For this reason, the Newey and West (1987) robust standard errors are utilised to ensure the fitness of regression analysis. Further diagnostic procedures, including the ARCH-LM test, were applied to ensure that all the ARCH effects are fully modelled in the BEKK-MGARCH model (Eqn (6)).

6.1 TA profitability assessment based on dynamic CAPM

Based on a test of the 24 TTRs (listed in Table A1, in Appendix) using the dynamic CAPM model, the results of each exchange rate (BTC/AUD, BTC/EUR, BTC/JPY and BTC/ZAR) are reported separately in Tables 2–5, respectively.

6.1.1 BTC/AUD exchange rate. In what follows, we report on TA performance based on the returns of moving average rules applied on conditional CAPM for BTC/AUD exchange rate.

Based on BBL's mean difference test, the results in Table 2 compare TA generated trades to those of buy–hold strategy. The average number of buy and sell signals is 68 and 175, respectively. The findings on 24 trading rules tested show that all buy signals are significantly different from the buy–hold strategy at 1% level, while just under 80% of the sell signals are significant at conventional levels. This means that while buy signals yield better results than sell signals, they both outperform the buy–hold strategy. There are other noteworthy factors to observe in the results.

The trading rule that stands out is trading rule number 10, which is designed as (2,50,1) meaning 2 short days, long-moving average of 50 days and a premium band of 1% (hereafter and for convenience trading rules will be cited as *TTR* No. *#*, as sequenced in the result tables). *TTR* No. 10 produces 223 sell signals with average return of -0.12% compared to the 20 buy signals with the highest average daily returns of 1.5%. These returns are statistically significant at 1% level. Further, all the indicated 20 buy signals produce positive returns, while only 44% of the 223 sell signals achieve returns greater than zero (columns 8 and 9). Other trading rules that produce relatively high buy returns which are significantly different from the sell returns are TTR No. 5 (5,20,0) and TTR No. 20 (1,200,1). Thus, contrary to Brock *et al.* (1992) who note that profits derived from trading rules depend on the number of signals, even if there are fewer buy signals, it is discovered that the buy return is higher than the sell return.

On aggregate, the standard deviations of buy signals (column 5) and neutral or buy-hold strategy (column 6) are comparably higher than the sell signals (column 7). This implies that while the buy trades are riskier relative to sell signals, these buy signals are no riskier than the market. The average fraction of buy and sell returns (columns 8 and 9), which are above zero, are 87 and 35%, respectively. The daily mean returns for buy, buy-hold and sell are 0.9%, -0.01% and -0.2%, respectively (columns 10, 11 and 12). The buy returns are

JCMS

| | | | | | | | | | | | | | | | | ., | | | | | | | | | | | |
|------|--------------------|---------|----------|-------------|----------|---------|----------|-------|----------|---------|----------|---------|---------|----------|-------------|---------|----------|---------|---------|---------|---------|---------|---------|---------|--------|-------|---|
| | (sell) | *** | ~ | *** | ~ | *** | | *** | * | *** (| ~ | *** | *** | *** | | * * | *** | *** | * | *** | *** | *** ~ | *** (| *** | * | | Technica |
| 16 | t stat(| 7.918 | 1.102 | 5.832 | 1.083 | 4.261 | -1.475 | 8.594 | 2.098 | 6.020 | 0.497 | -8.535 | -11.227 | 8.501 | 0.264 | 7.654 | 2.772 | 7.494 | 1.683 | 10.131 | 3.431 | 10.328 | 2.890 | 9.062 | 1.797 | | b d d d d d d d d d d d d d d d d d d d |
| | | * | * | * | ** | ** | * | ** | * | * | * | * | * | * | * • | * * | * | * * | * | * | * | ** | * | ** | * | | udes, ea buy ar standå |
| 15 | at(buy | 352 * | % 251 | * |)51 * | 517 * | * 66(| * 087 | 877 * | |)61 * | 285 * |)43 * | » 16(| 87 87 | 794 * | 926 * | 12 | 104 | * 80 | 349 * | * 600 | * 090 | 197 * | 392 * | | this ding rule age of through the sell age of |
| | t st | с. Г | -4.8 | -4.] | -3.0 | -4.6 | -4.(| -5.5 | -6- | -6.] | -8.(| -9.5 | -8.(| -7.(| -10.8 | -0- | 8 | -7- | -7.4 | -7. | -10.6 | -6.(| -11.(| -7.1 | -10.5 | | e 12 tra s create bercent hase a hase a |
| 14 | E(sell) | 001 | 100 | 100. | .001 | 100. | 100. | 100 | 100 | 100. | 100. | 100. | 100. | 100 | 100 | 100 | 100. | 100 | 100. | 100. | 100. | 100. | .001 | 100. | 100. | 100. | . The purc |
| | , S.F | 0 | 0 | 0 | 0 | 0 | 0 | 0 | 0 | Ö | 0 | Ö | Ö | 0 | 0. | 0 | Ö | 0 | Ö | 0 | Ö | Ö | 0 | 0 | 0 | Ö | fact, th ber of 3 and 7 12. Th 12. Th |
| 13 | S.E(buy | 0.001 | 0.002 | 0.001 | 0.003 | 0.001 | 0.003 | 0.001 | 0.001 | 0.001 | 0.002 | 0.001 | 0.002 | 0.001 | 0.001 | 0.001 | 0.001 | 0.001 | 0.002 | 0.001 | 0.001 | 0.001 | 0.001 | 0.001 | 0.001 | 0.001 | |
| 7 | lle | 004 | 100 | 003 | 000 | .003 | 000 | 005 | 100 | 004 | 001 | .005 | 007 | 900 | 001 | 002 | 002 | 002 | 100 | 000 | 002 | .005 | 002 | 005 | 002 | 002 | to greater to greater the trans 10, min 2. The second seco |
| | urns I Sc | 0 | 0 | 0 | 0 | 0 | 0 | 0 | 0 | 0 | 0 | 0 | ° | 0 | 0 | | | 0 | 0 | 0 | 0 | 0 | 0 | 0 | 0 | 0 | allest n colum shown colum colum cance J |
| 11 | lean reu Neutra | 0.001 | 0.000 | 0.000 | 0.000 | 0.000 | -0.001 | 0.000 | 0.001 | -0.001 | -0.001 | -0.001 | 0.000 | 0.00 | -0.001 | 0.000 | 0.000 | -0.001 | 0.000 | 0.001 | 0.000 | 0.001 | 0.000 | 0.000 | 0.000 | 0.000 | from sur te listed i uurus is a hown in te signific |
| 10 | Buy | 0.004 | 0.010 | 0.004 | 0.013 | 0.005 | 0.011 | 0.005 | 0.013 | 0.005 | 0.015 | 0.009 | 0.013 | 0.007 | 0.014 | 0.006 | 0.013 | 0.006 | 0.012 | 0.007 | 0.013 | 0.006 | 0.013 | 0.007 | 0.012 | 0.009 | olumn 1 ttions art d sell retr ms are s ms are s l as their l as their |
| 6 | %0 < | | 2 | Q Q | so So | 5 | 6 | 80 | so So | 0 | 4 | 6 | 4 | - | ري ب | 7 | - | 2 | 4 | | 5 2 | 7 | Ţ. | <u></u> | ŝ | 5 | tregula tranan un retu as wel |
| | sell : | ¢73 | 4 | ст э | 4 | ന | 4 | 2 | 4 | cr) | 4 | 2 | 4 | | 4 | | 4 | | 4 | - | 4 | - | 4 | CI | 4 | cr) | are liss in trading tistics tistics |
| | ×etur > 0% | 22 | 5 | 55 | 92 | 22 | 0 | 33 | 0 | 22 | 0 | 2 | Q | 22 | 2 | × × | 2 | 8 | Q | 4 | Q | 80 | 0 | 4 | 0 | 22 | ations rrious t and so e <i>t</i> -star |
| ~ | Buy : | Ŷ | . 0, | J | 0, | J | 1 | | 1 | | Ξ | | H | ~ | Ξ. | | H | | H | ~ | 1 | ~ | 1 | ~ | Ξ | ~ | g regul The va how th how th |
| 2 | Sell | 0.002 | 0.003 | 0.002 | 0.003 | 0.002 | 0.003 | 0.002 | 0.002 | 0.002 | 0.002 | 0.002 | 0.003 | 0.002 | 0.002 | 0.002 | 0.002 | 0.002 | 0.003 | 0.002 | 0.002 | 0.002 | 0.002 | 0.002 | 0.002 | 0.002 | trading band.' buy, i d 16 sl |
| | deviai itral | 08 | 60 | 908 | 600 | 60 | 60 | 008 | 60 | 08 | 908 | 908 | 80 | 60 | 10 | 60 | 60 | 60 | 60 | 60 | 60 | 08 | 60 | 08 | 60 | 60 | The 24 Ithe 24 |
| | ndard Neu | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | *10 *10 *10 |
| 5 | Buy | 600.0 | 0.007 | 0.00 | 0.011 | 0.00 | 0.008 | 0.008 | 0.006 | 0.008 | 0.008 | 0.008 | 0.006 | 0.007 | 0.006 | 0.007 | 0.007 | 0.007 | 0.008 | 0.007 | 0.006 | 0.007 | 0.006 | 0.008 | 0.006 | 0.007 | each 1 4. The ally, α * 5%, |
| 4 | N(sell) | 130 | 230 | 131 | 232 | 133 | 234 | 134 | 221 | 134 | 223 | 133 | 224 | 129 | 215 | 129 | 215 | 130 | 217 | 124 | 212 | 125 | 212 | 124 | 214 | 175 | umus in an which s a and s a and a a and t a s a a d a a a t ion t ion |
| | uy) 1 | 4 | ŝ | 2 | 2 | 0 | 8 | 0 | 2 | 0 | 0 | 0 | 6 | 4 | <u>80</u> - | 4 | 2Q | <u></u> | 9 | 6 | | 8 | H | 6 | 6 | 80 | 16 columns blumns isplay and tee: **** |
| 0.05 | ŊŊ | 11 | - | Ξ | - | 11 | | Ξ | | Ξ | C1 | Ξ | 1 | Ξ. | | Ξ | -1 | | -1 | Ξ | (1) | 11 | (7) | Ξ | C1 | 9 | reare her rej min α miniα hificar bwn cc |
| 2 | TTR | 20.0) | 20.1) | 20,0) | 20,1) | 5,20,0) | 5,20,1) | 50.0) | 50.1 | 2,50,0) | 2,50,1) | 5,50,0) | 5,50,1) | ,150,0) | (150,1) | ,150,0) | ,150,1) | ,150,0) | ,150,1) | ,200,0) | ,200,1) | ,200,0) | ,200,1) | ,200,0) | 200.1 | е | Table 2 BTC/AUD results for BTC/AUD results for |
| | ol | D | 2 | .57 | | <u></u> | <u> </u> | , C | | | 0 | 1 | 2 آ | 3 5 | 4 I | 2 | 6 | 7 | 8 | 6 (] | 0 (1. | 1 | 5 5 | 3 9 | 4 | Verag | BBL's mea difference tes |
| | 4 | | 2 | က | 4 | S | 9 | 0 | ∞ | 9 | Г | | - | - | - | | - | | - | | 2 | 2 | \sim | \sim | \sim | 4 | - KH RH & WWW |

| JCMS | | ([]; | * * * | * * | | * * | * 1 | * * | * * * | | * * * | * * * | * * * | | * * · | * 1 | • | 1 1 1 1 | -% | *** | | * * | | | ch is arch | gnals s are | 3 | | |
|---|-----------------|-------------|-------------|----------------------|----------|------------|----------|------------------|------------------------|----------|-------------|-------------|-------------|-----------|-----------|--------------|------------|------------------|--|------------|-----------|-----------|-----------|--------|----------------------------------|-----------------------------|-----------------------|--------------------|-------------|
| 6,1 | 16 | t-stat(se | 6.624 | 0.858 4.923 | -0.040 | 5.026 | 6.026 | 7.941 | 5.324 | 0.451 | -14.967 | -18.136 | 8.707 | -0.012 | 7.113 | 1.616 | 0.460 | 0.430 | 9.92U 1 E01 | 100.1 | 0.531 | 8.831 | 0.137 | | each of whi | and sell sig | | | |
| 16 | 15 | at(buy) | *** 96 | 92 *** 52 *** | *** 69 | 26 *** | 16 *** | *** /G | 52 *** | *** 26 | 89 *** | 72 *** | *** 62 | 80 | 81 *** | 80 *** 98 | TQ | 00 | 10 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 | *** 00 | 35 *** | *** 60 | 47 | | ling rules, 6 d throughc | age of buy . A sell stan | | | |
| | | t-sta | -5.0 | -6.7 | -3.3 | -3.7 | -1.4 | 9.0- 8.4- | -2.9 | -4.6 | -13.5 | -4.5 | -6.3 | -8.0 | -6.9 | -0 4. c | 4.0 | 0.0 | | 19.0 | 8.8- | -6.8 | -0.5 | | e 12 trac ls create | percenta chase an | | | |
| | 14 SF | (sell) | 0.001 | 0.000 | 0.001 | 0.001 | -3.114 | 0.000 | 00000 | 0.000 | 0.000 | 0.000 | 0.001 | 0.001 | 0.001 | 100.0 | 100.0 | 100.0 | 100.0 | 100.0 | 0000 | 0.001 | 0.001 | 0.001 | ct, there ar | and 7. The | | | |
| | $^{13}_{ m F}$ | (fud) | 0.001 | 0.002 | 0.004 | 0.001 | 0.000 | 0.000 | 0.001 | 0.002 | 0.001 | 0.002 | 0.001 | 0.002 | 0.001 | 0.001 | 100.0 | 200.0 | 100.0 | 10000 | 0.001 | 0.001 | 0.020 | 0.002 | ttest. In fa | 11 and 15 | 1 0100 | | |
| | 12 | Sell | -0.009 | -0.006 | -0.005 | -0.008 | 0.005 | -0.009 | -0.008 | -0.006 | 0.002 | 0.003 | -0.010 | -0.006 | -0.009 | -0.006 | -0.009 | c00.0- | 0T0'0- | 0000 | -0.006 | -0.010 | -0.006 | -0.007 | est to grea | wn in colt humus 10 | ce levels | | |
| | 11 ean rehim | Neutral | -0.005 | -0.005 | -0.005 | -0.005 | -0.005 | -00.00 200.00 | -0.006 | -0.006 | -0.006 | -0.005 | -0.005 | -0.006 | -0.006 | -0.005 | -0.000 | C00.0- | 200.0- | 2000- | -0.006 | -0.005 | -0.006 | -0.005 | rom small listed in c | urns is sho | significan | | |
| | 10 M | Buy | -0.001 | -0.006 | 0.009 | -0.003 | -0.006 | -0.001 | -0.002 | 0.004 | 0.005 | 0.006 | -0.001 | 0.007 | -0.001 | 0.006 | 100.0- | C00.0 | 100.0 | -00.00 | 0.007 | -0.001 | 0.005 | 0.003 | column 1 f ations are | nd sell retu | ll as their | | |
| | 6 0 ~ 4 | Sell > 0% | 8 | 8] 6: | 23 | 12 | 0.008 | 7 0 | 9 | 18 | 7 | 20 | 7 | 17 | CN ; | 19 | 0 | ہ م | 7 4 | oj cr | 16 | 2 | 20 | 12 | s are listed in trading regul | uy, neutral, ai | utistics as we | | |
| | 8 Retur | Buy $> 0\%$ | 39 | 001 55 | 100 | 34 | 22 | 47 87 87 | 3 8 | 73 | 41 | 88 | 39 | 100 | 41 | 100 100 | χ χ | χç | 747 | 100 1 | 92 92 | 47 | 88 | 99 | g regulations The various | viation for bu | how the <i>t</i> -sta | | |
| | 7 ation | Sell | 0.002 | 0.003 | 0.003 | 0.002 | 100 | 0.002 | 0.002 | 0.003 | 0.002 | 0.003 | 0.003 | 0.003 | 0.002 | 0.003 | 200.0 | 0.003 | c00.0 | 60000 | 0.003 | 0.003 | 0.015 | 0.003 | 24 tradin s a band. | he huv | and 16 s | | |
| | 6 Jard dav | Neutral | 0.006 | 0.006 | 0.007 | 0.007 | 0.003 | 0.006 | 0.006 | 0.006 | 0.006 | 0.006 | 0.007 | 0.007 | 0.007 | 0.007 | /00/0 | /00/0 | 100.0 | 0000 | 0.007 | 0.006 | 0.007 | 0.007 | uble. The ntroduce | olling sta | umns 15 | 10 | |
| | 5 Stan | Buy | 0.006 | 0.004 | 0.010 | 0.006 | 0.007 | 0.006 | 0.006 | 0.007 | 0.007 | 0.006 | 0.005 | 0.006 | 0.005 | 0.005 | C00.0 | CUU.0 | 200.0 | 0000 | 0.005 | 0.005 | 0.056 | 0.008 | n each Ta ch then i | 4. The r | nally, col | ** 5%, * | |
| | 4 ations | N(sell) | 126 | 236 126 | 238 | 127 | 0.008 | 129 231 | 132 | 232 | 132 | 235 | 129 | 228 | 128 | 228 | 621 | 231 | 071 | 195 195 | 230 | 124 | 235 | 180 | dumns ir tion, whi | uns 3 and | d 14. Fir | *** 1%, itation | ומרזחו |
| | 3 Observy | N(buy) | 117 | 7117 | 2 | 116 | 240 | 114 13 | 111 | 11 | 111 | 8 | 114 | 15 | 115 | 12 12 | CII CII | 71 C | 110 | 118 | 13 | 119 | 8 | 63 | e are 16 cc er replicat | n in colum o is disnla | nns 13 an | uticance: | אזז בסוווסר |
| Table 3. BTC/EUR results for BBL's mean | 2 | TTR | (1, 20, 0) | (1,20,1) (2.20.0) | (2,20,1) | (5, 20, 0) | (5,20,1) | (1,50,0) | (1, 30, 1) (2.50.0) | (2,50,1) | (5, 50, 0) | (5, 50, 1) | (1, 150, 0) | (1,150,1) | (2,150,0) | (2,150,1) | (0,150,0) | (1,001,6) | (1,200,1) | (1,000,1) | (2.200.1) | (5,200,0) | (5,200.1) | rage | e(s): Ther ed to anoth | od is shown ar than zer | wn in colur | istical sign | 0. (s)on 1 |
| unierence test | - | No | | N 03 | 4 | ß | က၊ | α | ი ი | 10 | Ξ | 12 | 13 | 14 | 15 | 19 | L CF | 2 5 | <u>ا</u> م | 35 | 52 | 23 | 24 | Ave | Not link(| perio | shor | Stat | |

| | | * | | * | | * | | * | | * | | * | * | * | -1 | e | -1 | ÷ | -¥- | | -*- | | * | | | പ്രപ്രപ്ര പ്രപ്രപ്രം | • • |
|------------|------------------|----------------|--------|--------|----------|----------------|-------------|--------|----------|--------|---------|----------|----------|------------|-----------------|----------------|-----------------|------------|------------|------------|--------|--------|---------|--------|---------|--|-----------------------|
| 6 | (sell) | °** 80 | 0 | ** 6 | 6 | :** ** | 2 | :** 0 | * 0 | :** 2 | 7 | ÷**9 | ** ** | ** 9 1 | - | 0 | | 0 5 | ~ .(/ * | * °° | ** | 5° | ** 8 | 2 | | I echn trading ru | ica les |
| 1 | t stat | 7.08 | 1.12 | 6.24 | 0.41 | 5.81 | 0.28 | 9.30 | 1.69 | 6.59 | 0.13 | -13.33 | -13.66 | 8.62 | -1.25 | 27.2 | 1.00 | 0.70 | 10.001 | 2.08 | 11.99 | 2.05 | 8.79 | 0.15 | | profitabi | lity |
| | | * | *** | *** | *** | *** | *** | *** | *** | * | *** | * * | * | * - | * 1 | () () | 6 X | ** | *** | * | *** | *** | **** | | | ules, ez bughou I stand I stand | |
| 15 | tat(buy | , 092 | 620 | 359 | 390 | 302 | 722 | 451 | 452 | 020 | 312 | 003 | 267 | 318 8 1 | . 1 19 12 | 210 | 170 | 710 | 511 × | 333 | 164 | 722 | 325 | 633 | | ading r ted thru and sel | 17 |
| | t s' | - | -2 | -4- | | | -2 | -2. | Ö | -0- | -5 | -12. | -0. | -7- | xi t | - | ġ t | ; | τ | -10^{-1} | 9 | 0 | -7. | -0 | | ls crea percer chase a | |
| 14 5 F | (sell) | 0.000 | 0.000 | 0.000 | 0.000 | 0.001 | 0.000 | 0.000 | 0.001 | 0.000 | 0.000 | 0.000 | 0.000 | 0.001 | 0.001 | 100.0 | 100.0 | | | 0.001 | 0.001 | 0.001 | 0.001 | 0.001 | 0.001 | there ar of signation of the pur | |
| 13 13 | buy) | 100.C | 00.0 | 100.C | 1000.C | 100.C | 700.C | 100.C | 100.C | 100.C | 0.002 | 100.C | 0.002 | 100.0 | 2007 | 100.0 | 2007 | | 200.0 | 1000 | 1001 | 0.002 | D.001 | 0.016 | 0.002 | n fact, t mber c od 12. J | |
| | 1 | 900 | 90 | 900 | 80 | 900 | 80 | 200 | 50(| 200 | 500 | 33 | 20 | 8 | 50 | 8 | 55 | žŽ | | 84 | 8 | 20 | 800 | 004 | 005 | The numeratest. I at the numeratest of the numer | |
| 12 | Se | -0.(| -0. | -0.(| -0.(| -0.(| -0.(| -0.(| -0.(| -0.(| -0.(| 0.0 | <u>.</u> | 0.0 | 0.0 | 0.0 | 0 | | | 0.0 | -0.0 | -0.0 | -0.0 | -0.(| -0.0 | to gree nmn 2.' nmns 10 levels | |
| 11 | eutral | -0.003 | -0.003 | -0.003 | -0.003 | -0.003 | -0.003 | -0.003 | -0.003 | -0.003 | -0.004 | -0.003 | -0.003 | -0.003 | -0.004 | -0.003 | -0.003 | -0.004 | | -0.003 | -0.002 | -0.003 | -0.003 | -0.003 | -0.003 | nallest in colu n colu icance ïcance | |
| Moo | N NG | - 00 | - 20 | - 00 | - = | - 00 | 15 | - 10 | - 20 | - 10 | - 80 |)5)5 | 10 | 23 | 10 | 22 | י <u>2</u> צ | 2 2 | , २ २ | | - 2 | 10 | 22 | - 20 | -)5 | rom sr elisted hown i signif | |
| 10 | Buy | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | -0.0 | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | | | 0.0 | 00 | 0.0 | 0.0 | 0.0 | 0.0 | umn 1 f ons are sell ret is their is their | |
| 6 | > 0% | × | | 0 | <u>ი</u> | 2 | | 0 | 2 | 2 | | 2 | | 69 | ο, ι | ں ، | | 0 G | 2 LC | n oc | i cr. | 6 | 6 | | | in colu egulati return s well a | |
| / | Sell | - | (1) | ~ | (1) | - | (1) | - | (1) | - | (1) | - | (1) | (| -1 | C | -, Ci. | - 0 | J | ~ | 1 | 51 | | (1) | 21 | e listed ading r neutrz l mean stics as | |
| 2 Dotti | > 0% | 1 8 | 88 | 51 | 8 | 1 9 | 0 | 80 | 10 | 26 | 00 | 22 | 8 | 5 | 25 | n N | <u></u> 22 5 | ខ្ល | 20 | 22 | 22 | 18 | 33 | 92 | 73 | ons art our buyz <i>t</i> -statis | |
| ~ | Buy | 7 | ~ | | 1 | 7 | H | | | | H | | H | | H, | | | | | , 1 | | Ä | Ŭ | 0. | | egulati he vari- ation fi ation fi w the | |
| 7 | Sell | 0.002 | 0.002 | 0.002 | 0.002 | 0.002 | 0.002 | 0.002 | 0.002 | 0.002 | 0.002 | 0.002 | 0.002 | 0.002 | 0.002 | 200.0 | 0.002 | 0000 | 2000 | 0.002 | 0.002 | 0.002 | 0.002 | 0.013 | 0.003 | ading ra and TJ auy, ne Juy, ne J6 sho | |
| domint- | ueviai tral | 90 | 20 | 90 | 07 | 07 | 07 | 90 | 20 | 90 | 90 | 90 | 02 | 20 | 20 | 70 | 25 | 55 | 58 | 86 | 07 | .0 | 20 | 07 | 20 | e 24 trz ces a b 15 and 15 and | |
| 6 Gard | Neu | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | | | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | 0.0 | ale. Thus a number of the second seco | |
| 5 Ctro | Buy | 0.006 | 0.011 | 0.007 | 0.010 | 0.007 | 0.012 | 0.006 | 0.006 | 0.006 | 0.008 | 0.006 | 0.007 | 0.006 | 0.006 | 0.000 | 0.008 | 200.0 | 0000 | 0.005 | 0.005 | 0.007 | 0.006 | 0.055 | 0.009 | the Tath The ru unmus & Unmus & 5%, * 5%, | |
| 4 | V(sell) | 127 | 235 | 128 | 237 | 129 | 240 | 129 | 229 | 131 | 230 | 130 | 233 | 125 | 228 | 07.T | 877 | 071 730 | 199 | 227 | 122 | 228 | 120 | 232 | 179 | nns in 'which' which' which' and 4 din col di in col 1%, **ina ion | |
| tornoo | serval. 1y) A | .0 | ~ | 10 | 0 | 4 | ~ | Ŧ | 6 | ~1 | ~ | ŝ | 0 | 201 | 1 01 | ~ 1 | 10 | - 0 | ۰ - | - ~ | . – | | ~ | ~1 | 6 | 6 colur ication splaye ce: *** nputat | |
| 0.5 | N(br | 11(| | 11 | - | Π | | Π | 12 | Ξ | ÷ | Ξ | Ä | Ë | ÷ ; | Ì | -ĭ È | = - | 4 C F | 11 | 12 | i =i | 12 | Ц | 9 | e are 1 er repl n in co o is dis mns 13 mns 13 wn cor wn cor | |
| 2 | TTR | 20,0) | 20,1) | ,20,0) | ,20,1) | ,20,0) | ,20,1) | 50,0) | 50,1) | (50,0) | (50,1) | ,50,0) | 50,1) | 150,0) | 150,1) | (0,061 | (T,0,1) | 1501) | (T,00) | 200.1) | 2000 | 200,1) | 200,0) | 200.1) | | Tak s anothor s anothor s anothor BTC/IPY result BTC/IPY result | le 4 ts for |
| | 1 | []. | Ľ, | СĴ | СĴ | <u>(</u> 2) | <u>(</u> 2) | Ĺ. | Ĺ, | C) | С) С | (2) | ю́. | | d é | সূর্ | সূদ | ה ע נ | 20 | | | 10 | 20 | 2 | verage | BBL's BBL's BBL's BBL's BBL's | near e tes |
| | ž | | \sim | က | 4 | വ | 9 | ⊳ | ∞ | 6 | 10 | Ξ | 12 | Ξ, | 7,1 | <u>, , ,</u> | 16 | 1 8 | 10 | 20 | í R | 2 | 23 | 24 | A | N SS SP E E N | |

| JCMS | | (IIs | * * | | * * * | 1 1 1 | | *** | | *** | | * * * | * * * | * * * | | * * * | a series a s | 66 | -11 -11 -11 | * | * * | | *** | | | ich is | encin znals | sare | | | |
|---|--------|-----------------------|----------|----------|-----------------|-------------------|------------------|----------|----------|-------------|-------------|-------------|-------------|-------------|-------------|-------------|---|-------------------|-------------------|-----------|-----------|-------------|-------------|-------------|--------|----------------|----------------------------------|----------------|--------------------------------|-------------|--|
| 0,1 | 16 | t-stat(se | 6128 | 0.170 | 5.257 | 0.173 0.173 | 4.710 0.947 | 7.260 | 1.086 | 6.459 | -0.975 | -18.851 | -22.039 | 8.352 | -1.126 | 7.106 | 0.706 | 4./99 | 16011- | 1.134 | 9.705 | 0.159 | 7.824 | 0.226 | | each of wh | and sell sig | idard error | | | |
| 10 | | (Ar | * * | * * | * * * | 6 N 6 N 6 N | | *** | *** | * * * | * * * | * * * | * * * | * * * | * * * | * 1 | * 1 | 6 N 6 N 6 N | 1 | ** | *** | * * * | * * * | * * * | | rules, | of buy | ell star | | | |
| 10 | 15 | <i>t</i> -stat(bu | -3 997 | -9.034 | -5.030 | -3./95 | -4.205 -0.871 | -5.349 | -6.202 | -6.215 | -4.077 | -16.896 | -14.395 | -6.423 | -8.144 | -6.445 | -8.562 | -01/10 010 k | 210.9- | 19 574 | -5.286 | -10.178 | -4.480 | -4.045 | | 12 trading | s createu u percentage | chase and s | | | |
| | 14 | S.E (sell) | 0.001 | 0.001 | 0.001 | 100.0 | 100.0 | 0.001 | 0.001 | 0.001 | 0.001 | 0.001 | 0.001 | 0.001 | 0.001 | 0.001 | 0.001 | 100.0 | 100.0 | 100.0 | 0.001 | 0.001 | 0.001 | 0.001 | 0.001 | there are | I signal 7. The | The pure | | | |
| | 13 | S.E (buy) | 100.0 | 0.001 | 0.001 | 0.003 | 0.015 | 0.001 | 0.002 | 0.001 | 0.002 | 0.001 | 0.001 | 0.001 | 0.001 | 0.001 | 100.0 | 100.0 | 200.0 | | 100.0 | 0.001 | 0.001 | 0.002 | 0.002 | In fact, | 5, 6, and | and 12. | | | |
| | 12 | s Sell | -0.010 | -0.007 | -0.010 | -0.00/ | 010'0- | -0.011 | -0.008 | -0.011 | -0.007 | 0.002 | 0.004 | -0.012 | -0.008 | -0.011 | -0.008 | -0.011 | 100.0- | 710.0 | -0.012 | -0.008 | -0.011 | -0.008 | -0.008 |) greatest. | n columns | ns 10, 11, a | evels | | |
| | 11 | ean return Neutral | -0.007 | -0.007 | -0.008 | /00/0- | -0.008 | -0.007 | -0.007 | -0.008 | -0.008 | -0.008 | -0.007 | -0.007 | -0.008 | -0.007 | -0.007 | -0.008 | 100.0- | 100.0- | -0.007 | -0.007 | -0.007 | -0.008 | -0.007 | smallest to | eu m colun is shown i | n in columi | nificance le | | |
| | 10 | M. Buy | -0.004 | 0.004 | -0.004 | C00.0 | -0.004 | -0.003 | 0.005 | -0.003 | -0.001 | 0.004 | 0.002 | -0.003 | 0.004 | -0.003 | 0.002 | -0.003 | 100.0 | c00.0- | -0.003 | 0.002 | -0.004 | 0.001 | 0.000 | mn 1 from | ell returns | are shown | s their sigi | | |
| | 6 | n > 0 Sell > 0% | 4 | 15 | က် | c۱ ۲ | 16 | 2 | 14 | co | 14 | 4 | 15 | 1 | × | က္ | 12 | 0[| ۲۲ ۲ | 1 ں | 1 cc | 14 | က | 6 | 8 | listed in colu | uing regulation neutral, and s | mean returns | tics as well a | | |
| | 8 | Return Buy > 0% | 24 | 100 | $\frac{26}{20}$ | 00 | C 12 | 8 8 | 80 | 32 | 43 | 25 | 80 | 27 | 12 | 33 | 82 | 17 | 00 | 071 | 23 | 78 | 23 | 83 | 50 | gulations are | ie various tra ation for buy, | utral and sell | w the <i>t</i> -statis | | |
| | 7 | ttion Sell | 0.003 | 0.003 | 0.003 | 0.003 | 0.003 | 0.003 | 0.003 | 0.003 | 0.003 | 0.003 | 0.003 | 0.003 | 0.003 | 0.003 | 0.003 | 0.003 | 500.0 | 0.003 | 0.003 | 0.003 | 0.003 | 0.003 | 0.003 | rading re | uanu. 11 ard devia | buy, nei | d 16 Sho | | |
| | 9 | lard devia Neutral | 0.007 | 0.007 | 0.007 | 100.0 | 0.007 | 0.006 | 0.007 | 0.006 | 0.007 | 0.006 | 0.007 | 0.007 | 0.008 | 0.007 | 0.007 | 100.0 | 0.007 | 0000 | 0.007 | 0.007 | 0.007 | 0.007 | 0.007 | . The 24 th | rouuces a ing stand: | nd 9. The | nns 15 an) | | |
| | 5 | Stand Buy | 0.006 | 0.003 | 0.006 | 0.007 | 0.020 | 0.006 | 0.006 | 0.006 | 0.004 | 0.006 | 0.001 | 0.005 | 0.005 | 0.005 | 0.003 | 0.000 | 1000 | 0000 | 0.005 | 0.002 | 0.006 | 0.005 | 0.006 | the Table | 1. The roll | lumns 8 a | ally, colur * 5%. *1(| | |
| | 4 | ations N(sell) | 123 | 236 | 127 | 238 | 161 | 128 | 233 | 131 | 236 | 134 | 241 | 124 | 229 | 125 | 234 | 124 | 007 | 021 | 120 | 234 | 119 | 237 | 181 | lumns in | on, winc is 3 and 4 | yed in co | 1 14. Fm ** 1%. * | tation | |
| | e S | Observ N(buy) | 120 | 7 | 116_{1} | ۍ د د 11 | 611 2 | 115 | 10 | 112 | 7 | 110 | വ | 119 | 14 | 118 | 6 | 170 | 0 6 5 | C21 11 | 123 | 6 | 124 | 9 | 63 | are 16 co. | in columr | is displa | uns 13 and ficance: *: | n comput | |
| Table 5. BTC/ZAR results for BBL's mean difference test | 2 | TTR | 01 20 0) | (1,20,1) | (2,20,0) | (Z,ZU,I) | (0,02,0) | (1.50.0) | (1,50,1) | (2,50,0) | (2,50,1) | (5,50,0) | (5,50,1) | (1, 150, 0) | (1,150,1) | (2,150,0) | (2,150,1) | (b,150,0) | (T'0CT'C) | (1,200,0) | (2.200.0) | (2,200,1) | (5,200,0) | (5,200.1) | erage | te(s): There | ted to anoune tod is shown | ger than zero | wn in colum tistical signii | urce(s): Öw | |
| | | No | _ | 0 | ი · | 4 6 | ი <i>ч</i> | ~ | 8 | 6 | 10 | Ξ | 12 | 13 | 14 | 12 | 16 | 10 10 | 9 6 | 200 | 32 | 22 | 33 | 24 | Av | No. | per | larg | Sta | So | |

substantially different and higher than both the buy-hold and sell returns. To summarise, the above indicate that if the TA method were applied to BTC/AUD during the period under investigation, the TTR would have yielded returns' profitability better than that of a buy-hold strategy.

6.1.2 BTC/EUR exchange rate. The results of moving average rules based on conditional CAPM for the BTC/EUR exchange rate are reported in Table 3 and are discussed next.

An observation of the results in the context of the BBL's mean difference test shows that TA may be profitably used in the BTC/EUR exchange rate. The evidence for this finding is that out of the 24 tested trading rules, 96% (column 15) of the buy signals are strongly significant at 1% level. In addition, the sell signals are less impactful than buys, but 63% (column 16) of the executed experimental sell trades are significant. Furthermore, an overview of the rest of the table reveals other interesting insights. There are 180 sell signals compared to 63 buy signals, yet the buy trades outperform the sell (columns 3 and 4). For instance, a glance at columns 8 and 9 shows that there are significantly more returns that are greater than zero for the buy signals (averaging 66%) compared to the sell signals with an average of only 12%. Another example, TTR No. 6 (5,20,1) stands out in that it produces the greatest number of sell signals (of 240 trades) and it yields only three buy signals. The average daily mean returns for buy, buy–hold and sell trades are 0.3%, -0.5% and -0.7%, respectively (columns 10, 11 and 12). Again, it is noticed that the buy signals produce the highest return. Overall, the results show that the TA method is equally successful when applied to BTC/EUR. Nevertheless, it is also useful to observe that the results of BTC/EUR are less strong than BTC/AUD.

6.1.3 BTC/JPY exchange rate. The moving average rules results applied to conditional CAPM for the BTC/JPY exchange rate are shown in Table 4 and explained next.

The results of the BBL's mean difference test show that the TA method may be profitably applied to the BTC/JPY exchange rate. The evidence of this finding is based on the observation that out of the 24 tested trading rules, 23 buy signals are strongly significant at 1% level (column 15), while 67% of the experimented sell trades are significant at conventional levels (column 16). The background details also tell an informative story. In the total of 248 experimental trades from buy and sell signals, only 28% were generated by buy signals, yet the buy trades performed distinctly better with average returns of 73% compared to 21% of sell signals (see columns 8 and 9).

According to the above evidence, the average number of buy and sell signals for BTC/JPY is 69 and 179, respectively (columns 3 and 4). Again TTR No. 6 stands out as the TTR with the least buy signals but highest number of sell signals. Buy, sell and buy–hold have average standard deviations of 0.009, 0.003 and 0.007 (columns 5, 6 and 7) and daily mean returns of 0.5, -0.3 - 0.5% respectively. Therefore, the buy signal is more profitable than the sell and buy–hold strategy, but at slightly higher risk. Overall, the results support the efficacy of the TA method.

6.1.4 BTC/ZAR exchange rate. The findings of moving average rules based on conditional CAPM for the BTC/ZAR exchange rate are displayed in Table 5 and are discussed next.

Unlike Bitcoin exchange rates discussed above, BBL's mean difference test shows that TTR signals are individually significant at conventional levels but cannot be called profitable. The last row of the table shows that in the total trades of 244 (63 + 181) from the 24 trading rules, the buy, buy–hold strategy and sell trades yield an average standard deviation of 0.006, 0.007 and 0.003, compared to mean returns of 0, -0.7 and -0.8%, respectively. This implies that while the risk measure is comparable to other Bitcoin exchange rates, the mean returns are eliminated.

From the findings above (in Tables 2–5), it can be observed that the efficacy of TA is emphasised more in economies that appear to be cryptocurrency-friendly or have positively explored the market with adoption possibilities into mainstream finance (BTC/AUD and BTC/JPY). Evidence of diminishing profits is found in BTC/EUR and BTC/ZAR. The results

Technical trading rules' profitability

associated with BTC/EUR may be because of Bitcoin's becoming more efficient. The profits are entirely erased in the case of BTC/ZAR, where the official national position on cryptocurrency is rather unreceptive. Another reason why TA fails to yield profitable performance in the South African Bitcoin market (BTC/ZAR) is that the market itself is underdeveloped. Some believe that if there is influx of uninformed technical traders into a market, this may derail the technical strategy's power. The study's findings reveal that sell signals are more volatile than buy signals, which provide evidence that profits derived from *TTR* are not only compensation for risk.

6.2 Results of Kho's mean-spread test

The results of Kho's mean-spread test based on Eqn (8) are presented in Tables 6 and 7. These tables provide a report on the average spread between the conditional buy and sell signals along with their respective *t*-statistics. The tests in these tables are conducted under the null hypothesis that the spreads are not statistically different from zero. The *t*-statistics are computed using Newey and West's (1987) robust standard errors.

6.2.1 BTC/AUD and BTC/EUR exchange rate. Table 6 reports the results of Kho's buy– sell mean-spread test. The first two columns are tabulation and definition of rules. The first panel (columns 3 to 7) displays the results for BTC/AUD, while the second panel (columns 8 to 12) shows results for BTC/EUR.

The results of Kho's buy–sell mean-spread test provide enough evidence to reject the null hypothesis [H2] that the buy–sell spread is not different from zero, and we conclude that the buy signals differ from sell signals. Based on the 24 moving average trading rules that are tested, 54% of BTC/AUD are statistically significant at conventional levels, while all the *TTR*'s for BTC/EUR are strongly significant at 1% level. The rest of the table provides other supportive information. Eqn (8) is a special test equation (not conventional regression analysis) and as such the R^2 's are deemed reasonable, and the model is validated with a strongly significant *F*-statistics test. The above explanation supports the conclusion that the application of the TA method on BTC/AUD is profitable.

6.2.2 BTC/JPY and BTC/ZAR exchange rate. Table 7 reports the results of Kho's buy–sell mean-spread test for BTC/JPY and BTC/ZAR. The first panel (columns 3 to 7) displays the results for BTC/AUD, while the second panel (columns 8 to 12) is for BTC/EUR.

The results of the test equation for BTC/JPY (column 6) and BTC/ZAR (column 11) provide convincing evidence to reject the null hypothesis (H2) that the buy–sell spread of MA signals are not different from zero and we conclude that they are different. This provides evidence to submit that TA methods may be employed profitably in BTC/JPY and BTC/ZAR. As in Table 6, the regression analysis was generated with robust standard errors and subjected to similar validation procedures.

So far, the study has examined the profitability of TTRs under a dynamic CAPM model that accounts for market risk using CAPM-based equilibrium returns. The results show that a time-varying risk premium can explain profits. When comparing the mean returns between the equilibrium model and actual series, the study finds that after accounting for risk, the previously discovered profits using the actual series diminish, implying that the Bitcoin market has elements of time-varying risk premium. After accounting for risk, TA has been found profitable in BTC/AUD and BTCJPY, but such profits diminish when studying BTC/EUR and are completely gone in BTC/ZAR.

6.3 Robust and completeness test

Tables 8 and 9 present the results on the momentum effect being tested for the third study hypothesis [H3]. The hypothesis examines whether price changes (or returns) have an association with TA signals of buy and sell trades as per Eqn (9).

20

JCMS

| | | | | BTC/AI | 6 | | | , | | BTC/F | SUR | | 1 |
|-------------------|--|----------------------------------|---------------------------------|-------------------------------------|----------------------------------|-----------------|---------------|----------------|----------------|----------------|---------------------|--------------|-------------|
| $_{ m No}^{ m 1}$ | $_{TTR}^2$ | 3 Coeff | $^{4}_{ m SE}$ | 5 F-Stat | 6 <i>t</i> -stat | | R^2 | 8 Coeff | 9 S.E | 10 F-Stat | 11 <i>t</i> -sta | ıt | R^2 |
| | (1.20.0) | -0.003 | 0.095 | 0.002 | -0.026 | | 0.000 | 0.191 | 0.058 | 13.937 | 3.293 | * * | 0.006 |
| 2 | (1.20,1) | 0.816 | 0.063 | 168.093 | 12.964 | *** | 0.065 | 0.581 | 0.038 | 136.616 | 15.262 | ** | 0.053 |
| ı ი: | (2.20.0) | -0.033 | 0.113 | 0.266 | -0.291 | | 0.000 | 0.174 | 0.057 | 11.538 | 3.074 | *** | 0.005 |
| 4 | (2.20.1) | 0.692 | 0.120 | 124.588 | 5.740 | *** | 0.049 | 0.526 | 0.070 | 102.876 | 7.477 | *** | 0.041 |
| 2 | (5, 20, 0) | -0.017 | 0.100 | 0.069 | -0.167 | | 0.000 | 0.221 | 0.057 | 19.434 | 3.900 | *** | 0.008 |
| 9 | (5, 20, 1) | 0.893 | 0.064 | 217.271 | 13.868 | *** | 0.082 | 0.603 | 0.043 | 145.939 | 14.134 | *** | 0.057 |
| 7 | (1, 50, 0) | -0.009 | 0.113 | 0.020 | -0.081 | | 0.000 | 0.269 | 0.048 | 28.269 | 5.556 | *** | 0.012 |
| 8 | (1, 50, 1) | 0.516 | 0.104 | 63.091 | 4.983 | *** | 0.025 | 0.472 | 0.065 | 86.335 | 7.260 | *** | 0.034 |
| 6 | (2, 50, 0) | -0.012 | 0.093 | 0.036 | -0.134 | | 0.000 | 0.228 | 0.051 | 19.231 | 4.438 | *** | 0.008 |
| 10 | (2, 50, 1) | 0.449 | 0.144 | 50.641 | 3.122 | *** | 0.020 | 0.545 | 0.043 | 121.394 | 12.610 | *** | 0.048 |
| 11 | (5, 50, 0) | -0.041 | 0.107 | 0.378 | -0.380 | | 0.000 | 0.203 | 0.066 | 17.200 | 3.097 | * * | 0.007 |
| 12 | (5,50,1) | 0.627 | 0.101 | 102.019 | 6.194 | *** | 0.040 | 0.549 | 0.049 | 110.672 | 11.286 | *** | 0.044 |
| 13 | (1, 150, 0) | -0.039 | 0.098 | 0.000 | -0.392 | | 0.000 | 0.359 | 0.040 | 52.013 | 8.933 | *** | 0.021 |
| 14 | (1, 150, 1) | 0.342 | 0.115 | 28.206 | 2.982 | *** | 0.012 | 0.449 | 0.064 | 78.726 | 7.024 | *** | 0.032 |
| 15 | (2, 150, 0) | -0.033 | 0.100 | 0.241 | -0.328 | | 0.000 | 0.300 | 0.047 | 34.524 | 6.374 | *** | 0.014 |
| 16 | (2, 150, 1) | 0.381 | 0.137 | 35.254 | 2.782 | *** | 0.014 | 0.515 | 0.044 | 105.595 | 11.710 | ** | 0.042 |
| 17 | (5, 150, 0) | 0.034 | 0.094 | 0.278 | 0.363 | | 0.000 | 0.342 | 0.045 | 45.839 | 7.654 | *** | 0.019 |
| 18 | (5, 150, 1) | 0.504 | 0.108 | 69.646 | 4.672 | * * | 0.028 | 0.534 | 0.049 | 103.748 | 10.803 | *** | 0.041 |
| 19 | (1,200,0) | -0.122 | 0.080 | 3.222 | -1.530 | | 0.001 | 0.350 | 0.040 | 49.736 | 8.842 | *** | 0.020 |
| 20 | (1,200,1) | 0.325 | 0.116 | 25.094 | 2.815 | *** | 0.010 | 0.419 | 0.064 | 68.482 | 6.542 | *** | 0.028 |
| 21 | (2,200,0) | -0.050 | 0.086 | 0.517 | -0.578 | | 0.000 | 0.258 | 0.063 | 24.724 | 4.075 | *** | 0.010 |
| 22 | (2, 200, 1) | 0.357 | 0.114 | 30.879 | 3.131 | *** | 0.013 | 0.447 | 0.044 | 70.776 | 10.062 | *** | 0.028 |
| 23 | (5,200,0) | -0.170 | 0.088 | 6.969 | -1.919 | * | 0.003 | 0.327 | 0.045 | 42.097 | 7.339 | * * | 0.017 |
| 24 | (5, 200.1) | 0.345 | 0.098 | 25.905 | 3.538 | *** | 0.011 | 0.577 | 0.050 | 132.239 | 11.500 | *** | 0.052 |
| Note(s): | : The first coli | umn is the tab | ulation of n | noving average | s rules. Colum | n 2 is the 1 | technical tra | ading rule (| TTR) structu | re. For report | ing purpose, | the table is | divided |
| into two | panels. The fi | irst panel (colt | umns $3 \text{ to } 6$ |) displays the r | esults for BTC | /AUD and | d the second | d panel (colu | 12 mms 8 to 12 | gives results | for BTC/EU | R. Each ro | w (1–24) |
| presents | a regression ($\Delta II + h_{\Theta} E$ | output tor Eqn state are stat | n (8). Columr Tetically eige | 13 is the coeffic nificant at 1% | tent (α_{-1}) of X_{-1} | $_{-}(t-1)$ who | ere the inter | cept is set to | o zero. Other | regression sta | utistics are rej | ported in th | le rest of |
| Statistica | al significance | -stats at c stat | 5%, *10 | mmrant at 1 /0 | 10,01 | | | | | | | | |
| Source(| (s): Own com | putation | | | | | | | | | | | |
| | | | | | | | | | | | | | |
| _ | | | | | | | | | | | | | |
| sp AU | Kho | | | | | | | | | | | | tr |
| леас ЛD а | o's b | | | | | | | | | | | pro |] ad |
| l Tes nd B | , 1V—S | | | | | | | | | | |)11ta | Гес ing |
| st – H TC/ | Fab l | | | | | | | | | | | abi | hni ru |
| AUD | e 6. | | | | | | | | | | 21 | ity | cal les' |
| | | | | | | | | | | | • | | |

| JCMS 6,1 | , c | R^2 | 0.013 | 0.040 | 0.035 | 0.015 | 0.043 | 0.011 | 0.019 | 0:050 | 0.016 | 0.051 | 0.027 | 0.030 | 0.026 | 0.046 0.016 | 0.050 | 0.026 | 0.038 | 0.031 | 0.049 | 0.021 | 0.043 | divided v (1–24) e rest of |
|------------------------------------|--------|---------------------|------------|------------------|------------|------------|---------------|------------------|------------------|------------|------------|----------|-------------|-------------|-------------------------|-----------------------|-------------|-----------|-----------|-----------|-----------|-------------|-----------|--|
| | | t | * * | * * | *** | *** | * * | * * | * * | *** | *** | *** | *** | * * | 6 - 9 6 - 9 6 - 9 | * * | *** | *** | **** | ** | * * | * * * | ** | he table is R. Each rov oorted in th |
| 22 | R I | 11 <i>t-</i> sta | 5.552 | 13.775 7.055 | 11.341 | 6.312 | 11.585 | 6.106 10.081 | 6.458 | 14.235 | 8.046 | 12.730 | 11.236 | 11.270 | 9.2./6 | 7.028 | 13.459 | 11.750 | 12.782 | 10.436 | 13.426 | 8.274 | 13.879 | g purpose, t or BTC/EUH stics are rep |
| | BTC/ZA | FStat | 30.821 | 101.423 40.096 | 87.558 | 35.646 | 108.007 | 26.341 95 798 | 60.120 46.113 | 126.376 | 38.832 | 129.811 | 67.727 | 74.638 | 63.732 116 671 | 110.0/1 38.553 | 126.442 | 65.399 | 96.700 | 76.357 | 123.756 | 52.084 | 107.496 | e. For reportin Gives results f egression stati |
| | c | s.E | 0.045 | 0.033 | 0.038 | 0.041 | 0.040 | 0.039 | 0.046 | 0.035 | 0.035 | 0.040 | 0.032 | 0.036 | 0.038 | 0.040 | 0.037 | 0.030 | 0.035 | 0.036 | 0.037 | 0.037 | 0.033 | <i>(TTR</i>) structur mns 8 to 12). (zero. Other r |
| | c | 8 Coeff | 0.248 | 0.459 | 0.435 | 0.259 | 0.461 | 0.236 | 0.296 | 0.496 | 0.282 | 0.513 | 0.362 | 0.404 | 0.349 | 0.482 | 0.499 | 0.351 | 0.446 | 0.373 | 0.493 | 0.310 | 0.461 | ading rule (J panel (colu cept is set to |
| | t | R^2 | 0.005 | 0.050 | 0.055 | 0.004 | 0.066 | 0.003 | 0.003 | 0.045 | 0.004 | 0.044 | 0.003 | 0.025 | c00.0 | 0.005 | 0.061 | 0.007 | 0.033 | 0.012 | 0.025 | 0.002 | 0.063 | technical tra 1 the second ere the inter ere the inter |
| | | t | * * | * * | *** | * | * * * * | * * | * | *** | * | ** | * | * * | N N N N N N | * * | ** | *** | *** | * * | ** | | ** | m 2 is the \Im/A UD and $\zeta_{-}(t-1)$ wh |
| | , A | o <i>t</i> -sta | 3.083 | 6.451 2.133 | 6.464 | 2.037 | -7.458 | 2.1087 2.108 | 2.439 | 5.841 | 2.186 | 6.087 | 1.886 | 4.474 | 2.681 | 4.208 3.335 | 10.879 | 3.045 | 7.804 | 4.629 | 4.299 | 1.264 | 12.649 | s rules. Colur sults for BTC ent (α_{-1}) of λ level |
| | BTC/JP | 5 F Stat | 13.041 | 126.818 8.036 | 141.736 | 8.840 | 171.759 | 8.386 8.386 | 7.947 | 113.815 | 10.173 | 110.571 | 7.417 | 61.675 | 12.606 61 228 | 01.338 11.814 | 157.275 | 16.658 | 82.970 | 28.811 | 62.201 | 4.032 | 163.304 | oving average displays the re 3 is the coeffici lificant at 1%] |
| | | $_{\rm SE}^4$ | 0.070 | 0.102 0.078 | 0.105 | 0.085 | 0.096 | 0.086 | 0.072 | 0.104 | 0.086 | 0.101 | 0.085 | 0.109 | 0.079 | 0.064 | 0.064 | 0.080 | 0.068 | 0.069 | 0.110 | 0.093 | 0.057 | ulation of m mns 3 to 6) (8).Column stically sigr (%, *10 |
| | c | з Coeff | 0.217 | 0.656 0.167 | 0.676 | 0.173 | -0.719 | 0.140 | 0.174 | 0.609 | 0.188 | 0.617 | 0.161 | 0.488 | 0.210 | 0.408 | 0.699 | 0.244 | 0.533 | 0.320 | 0.475 | 0.118 | 0.718 | umn is the taburst panel (colu output for Eqn output are stati **** 1%, ** 5 putation |
| Table 7. Kho's buy-sell mean- | d | TTR^{2} | (1, 20, 0) | (1,20,1) | (2,20.1) | (5, 20, 0) | (5, 20, 1) | (L, 50, 0) | (2, 50, 0) | (2, 50, 1) | (5, 50, 0) | (5,50,1) | (1, 150, 0) | (1, 150, 1) | (2,150,0) | (2,150,1) $(5,150,0)$ | (5, 150, 1) | (1,200,0) | (1,200,1) | (2,200,0) | (2,200,1) | (5,200,0) | (5,200.1) |): The first col- panels. The fi panels. The fi s a regression c mns. All the F al significance (s): Own com |
| spread Test–BTC/JPY and BTC/ZAR | . | No | | C1 01 | 5 4 | വ | 91 | . 0 | 00 | 10 | 11 | 12 | 13 | 14 | 15 16 | 0T | 18 | 19 | 20 | 21 | 22 | 33 | 24 | Note(s into twc presents the colu Statistic Source |

Observation of the results for each of the regression outputs reported in Table 8 under both buy and sell signals provides strong evidence that there is an association between price changes of BTC/AUD and BTC/EUR, and the TA trade signals. This means that the null of H3 is rejected. The test serves to examine whether the properties of a trend apply to the study. In TA, the primary goal of price analysis (using a chart) is to detect the price trends and/or reversals in their early phases, which trigger trading.

The results of Table 9 provide an answer to H3 in respect of the relationship between price changes (or returns) and MA trade signals specifically for BTC/JPY and BTC/ZAR. The table shows that, like Table 8, there is convincing evidence that there is an association between the returns and trade signals. Overall, the study finds that concerning buy signals, the Bitcoin market possesses a momentum effect across the board as the results are strongly significant at conventional levels.

7. Discussion of results

To address the research question, 4 TA performance strategies were evaluated. To begin with, TTRs were applied to equilibrium returns calculated using the dynamic CAPM model. Second, the TA results were then replicated using the actual return series for comparison and robustness purposes. Thirdly, the study examined the extent to which time-varying risk premiums affect performance to determine the profitability of TA. Finally, the momentum effect in TA was investigated as a possible secondary mechanism to corroborate the results. The study's findings are similar to those of Gerritsen et al. (2020) and Grobys et al. (2020), who found that TTRs are more cost-effective than the buy-hold strategy when dealing with Bitcoin daily data. Their study investigates TA using trend-following tactics like the simple moving average. Further support is given by Tiwari et al. (2018) and Resta et al. (2020), who also emphasise the importance of TA and conclude that TA returns surpass those of a buvhold strategy. Overall, the study finds that the performance of the chosen TTR strategies is superior to the buy-hold strategy. The study is different in that it accounts for time-varying risk premiums and finds that some of the registered profits diminish after the factoring of risk. Bitcoin may be seen as a natural laboratory for using TA because it has no apparent fundamentals to investigate. The study offers evidence that the MA trading rule can predict Bitcoin returns even after accounting for time-varying risk premiums. Similar to known literature (Ülkü and Prodan, 2013), the profits found in Bitcoin exchange rates are found to diminish after accounting for time-varying risk premiums. In the evaluation of the momentum effect, the study finds similar evidence to that of Borgards (2021), in that the Bitcoin market possesses a momentum effect. Emerging markets are known to be less liquid than industrialised markets and have more concentrated trade. As a result, it is only natural to assume that inefficiencies could be more easily exploited in these markets. Gerritsen et al. (2021), however, find that the informational efficiency of Bitcoin has improved.

8. Conclusion

The paper examines the profitability of 24 TTRs in each of four Bitcoin exchange rates (BTC/AUD, BTC/EUR, BTC/JPY and BTC/ZAR). The results show that the chosen FMA TTRs are all successful in generating profitable signals for Bitcoin returns. The buy signals generate positive returns and sell signals generate negative returns, which are, on average, significantly different from the returns earned by the buy–hold strategy. The study results are consistent with Grobys *et al.* (2020) and Borgards and Czudaj (2021) who found that the trading rules are successful in predicting Bitcoin price movements. The study also observes that the Bitcoin market may possess a momentum effect. Though these results suggest market inefficiency, we find that time-varying risk premiums can diminish

Technical trading rules' profitability

| JCMS | | | * | * * | * * | * * | * * | | *** | *** | * * * * | *** | *** | *** | 4 4 4 | *** | *** | * * | * * | *** | *** | *** | (pən | 1 |
|---|-------|----------------------|-----------------------|-------------------|----------|-------------|-----------------------|-----------|-------------|------------|-----------------|----------------|----------|-------------|------------------|------------------|-------------|-------------------|----------------|-------------|-----------|-----------------------|---------|---|
| 6,1 | | 13 <i>t</i> -stat | -8.271 | -7.050 | -4.840 | -2.475 | -3.062 -3.616 | | -2.930 | -2.268 | -9.431 9.701 | -6.01 | -3.375 | -3.823 | 0000 | 0.832 -6.762 | -4.341 | -4.342 | -2.155 | -5.424 | | -7.489 | (contin | |
| 24 | | Sell 12 S.E | 0.003 | 0.004 | 0.003 | 0.005 | 0.003 0.002 | | 0.005 | 0.003 | 0.003 | 0.005 | 0.003 | 0.002 | 100.0 | 0.00.0 | 0.004 | 0.003 | 0.003 0.003 | 0.002 | 1000 | 0.003 | | |
| | UR | 11 Coefficient | -0.027 | -0.029 | -0.016 | -0.012 | -0.009 -0.006 | | -0.014 | -0.008 | -0.029 | -0.025 | -0.009 | -0.006 | 660.0 | -0.029 -0.029 | -0.018 | -0.015 | c10.0- | -0.008 | 0000 | -0.028 -0.022 | | |
| | BTC/E | | * * * | * * | * * | * * * | * * * * * * | | * * * | * * | + * + * | * | *** | * * * | 4 4 4 4 | * * | * * | * * | * * | * * * | *** | * * | | |
| | | 10 <i>t</i> -stat | 7.673 | 7.098 5116 | 5.350 | 3.580 | 3.513 - 13.967 | | 3.906 | 4.538 | 4.898 5 574 | 8.947 | 7.790 | -14.373 | | CC2.2- | 6.178 | 6.557 | 3 510 | -11.873 | | 0.934 7.824 | | |
| | | Buy 9 S.E | 0.004 | 0.005 | 0.006 | 0.008 | 0.005 0.001 | | 0.004 | 0.006 | 0.004 | 0.005 | 0.003 | 0.001 | 0000 | 0.005 | 0.004 | 0.005 | 0.000 | 0.001 | 1000 | 0.005 | | |
| | | 8 Coefficient | 0.029 | 0.037 | 0.032 | 0.030 | 0.018 - 0.018 | | 0.014 | 0.027 | 0.025 | 0.047 | 0.026 | -0.019 | | -0.024 | 0.023 | 0.034 | 0.014 | -0.015 | 100 0 | 0.036 | | |
| | | | ***** | * * | * | * * | * | | * * | * * * | + * * | * * | *** | * | -) -) | * * | **** | *) *) *) | * * | * | *** | * * | | |
| | | 7 <i>t</i> -stat | -7.509 | -1.781 6.912 | -2.064 | -5.471 | $0.316 \\ 1.766$ | | -3.559 | -7.248 | -1.898 9 000 | -6.448 | -2.574 | 2.242 | 001.0 | -2.192 | -9.158 | -4.926 | 907.6 | 1.595 | | -2.709 -8.609 | | |
| | | Sell 6 S.E | 0.001 | 0.000 | 0000 | 0.001 | 0.000 | | 0.000 | 0.001 | 0.000 | 100.0 | 0.000 | 0.001 | 0000 | 0000 | 0.001 | 0.000 | 100.0 | 0.001 | 0000 | 0.001 | | |
| | AUD | 5 Coefficient | -0.005 | -0.001 | -0.004 | -0.003 | 0.000 0.001 | | -0.001 | -0.004 | -0.001 | -0.004 | -0.001 | 0.001 | 100.0 | 10000- | -0.006 | -0.002 | 0.000 | 0.001 | 100.0 | -0.005 | | |
| | BTC/. | L. | * * * | * * * * * * | *** | * * * | * * * | | * * * | * * * * | + * + * | *** | *** | | 3 3 3 | * * | * * * | * * * * * * | + * + * | | *** | * * | | |
| | | 4 <i>t</i> -sta | 3.252 | 5.408 2 005 | 5.767 | 5.147 | 6.094 -1.074 | | 6.372 | 7.715 | 288.0 | 6.299 | 7.687 | -1.400 | | 0.233 9.533 | 8.269 | -9.426 | C04.7 | -0.922 | 0000 | 8.681 8.681 | | |
| | | Buy 3 S.E | 0.003 | 0.001 | 0.001 | 0.002 | 0.001 0.000 | | 0.001 | 0.001 | 100.0 | 0.001 | 0.001 | 0.000 | 0000 | 0.001 | 0.001 | 100.0 | 100.0 | 0.000 | 100 0 | 100.0 | | |
| Table 8. | | 2 Coefficient | <i>20 MA</i> 0.010 | 0.004 | 0.000 | 0.010 | -0.004 | 50 MA | 0.004 | 0.011 | 0.004 | 0.004 0.004 | 0.011 | -0.001 | 150 MA | C010.0 | 0.0118 | -0.0053 | 00000 | -0.004 | 200 MA | 010.0 | | |
| Momentum effect – BTC/AUD and BTC/EUR | | $\frac{1}{TTR}$ | Long-term (1,20,0) | (1,20,1) | (2.20.1) | (5,20,0) | (5,20,1) Intercept | Long-term | (1, 50, 0) | (1,50,1) | (2,50,0) | (5.50.0) | (5,50,1) | Intercept | Long-term | (1,150,1) | (2,150,0) | (2,150,1) | (5,150,0) | Intercept | Long-term | (1,200,1) $(1,200,1)$ | | |

| | at | * * * * * * * * * * * | . (2–7) nange | Technical |
|-------|----------------------|---|---|---------------|
| | 13 <i>t</i> -stz | -3.220 -5.253 -2.222 -1.768 -3.755 | irst panel each exch | profitability |
| Sell | 12 S.E | $\begin{array}{c} 0.006\\ 0.003\\ 0.005\\ 0.004\\ 0.002\end{array}$ | els. The f :ately for nd 200 | 25 |
| BUR | 11 Coefficient | -0.019 -0.018 -0.011 -0.001 -0.006 | two main pan eported sepai 20, 50, 150 a | |
| BTC/I | | * * * * * * * * * * * * * * * * * * * | ed into t ls are r amely, | |
| | 10 <i>t</i> -stat | 5.9826.6302.2034.493 -14.061 | ole is divide d sell signa age rules, n | |
| Buy | $^{9}_{ m SE}$ | $\begin{array}{c} 0.004\\ 0.005\\ 0.004\\ 0.007\\ 0.001\end{array}$ | en the tal or buy an ring aver | |
| | 8 Coefficient | $\begin{array}{c} 0.023\\ 0.034\\ 0.009\\ 0.031\\ -0.018\end{array}$ | ile (<i>TTR</i>). Th The results for retch for mov | |
| | | * * * * * * * * * * * * * * * * * * * | ding ru //EUR. erm sti | |
| | 7 <i>t</i> -stat | $\begin{array}{r} -9.055 \\ -2.042 \\ -3.410 \\ -10.479 \\ 2.464 \end{array}$ | cchnical tra lts for BTC each long-t | |
| Sell | 6 S.E | 100.0 0000.0 100.0 100.0 | for each te plays resu ced under | |
| /AUD | 5 Coefficient | -0.005 -0.001 -0.001 -0.006 0.001 | sion outputs nel (8–13) disj uts are report | |
| BTC | t | | regres ond pa n outpr | |
| | 4 <i>t</i> -sta | $11.564 \\ 7.058 \\ 11.076 \\ 9.769 \\ -2.113$ | the list of le the sec regressio *5%, *10 | |
| Buy | $_{\rm SE}^{3}$ | 0.001 0.001 0.001 0.001 0.000 | nn shows AUD, whi JR). The **1%, *: ttation | |
| | 2 Coefficient | $\begin{array}{c} 0.011\\ 0.004\\ 0.012\\ 0.006\\ -0.001\end{array}$ | The first colun ults for BTC// AUD, BTC/EI significance: * ': Own compu | |
| | $\frac{1}{TTR}$ | (2,200,0) (2,200,1) (5,200,0) (5,200,1) Intercept | Note(s): reports res rate (BTC/ Statistical Source(s) | Table 8. |

| JCMS | | | * * | * * | | * * * | *** | 44444 | * * * | | * * * | | * * | | -) | + * * | ** | | *** | * * * | *** | * | (pər | |
|--|-------|----------------------|-----------------------|------------------|----------|-------------------|-----------------------|--|-------------------|----------------------|-------------|-------------|-----------------|-----------|------------------|------------------|-------------|-----------|-----------------|-------------|-----------|-----------------------|---------|--|
| 6,1 | | 13 <i>t</i> -stat | -6.297 | -6.050 | 0.812 | -5.727 | -18.256 | 1000 | -6.265 | -0.480 | -7.524 | -1.030 | -7.727 | -180.91 | | 0450.1 -6.659 | -8.126 | -1.010 | -0.951 | -19.044 | | -9.450 -1.692 | (contin | |
| 26 | | Sell 12 S.E | 0.000 | 0.000 | 0.000 | 0.001 | 0.000 | | 0.000 | 0.000 | 0.000 | 0.000 | 0.000 | 0.000 | 0000 | 0.00.0 | 0.001 | 0.000 | 0.000 | 0.000 | 100.0 | 100.0 | | |
| | ZAR | 11 Coefficient | -0.003 | 0.000 - 0.003 | 0.000 | -0.003 | -0.007 | | -0.003 | 0000 | -0.004 | 0.000 | -0.003 | -0.00/ | 100.0 | -0.003 | -0.004 | 0.000 | 0.004 | -0.007 | 100 0 | -0.00 -0.001 | | |
| | BTC/2 | | *** | * * * * | *** | * * | * * | -to the t | *) *) *) | * * * * | | * * | * * | | 3 3 3 3 | * * | **** | * * | * * | * * * | *** | * | | |
| | | 10 <i>t</i> -stat | 6.436 | $8.180 \\ 4.234$ | 689.9 | 5.158 | -21.503 | | 8.073 | 4.7.768 | 0.915 | 6.978 | 7.494 | COR.12- | 1000 | -8.904 7.604 | 7.581 | 4.085 | 5.363 | -20.989 | | 6.498 6.498 | | |
| | | Buy 9 S.E | 0.000 | 0.001 | 0.000 | 0.001 | 00000 | | 0.000 | 0.002 | 0.003 | 0.001 | 0.001 | 0,000 | 0000 | 0000 | 0.000 | 0.002 | 0000 | 0.000 | 0000 | 0.001 | | |
| | | 8 Coefficient | 0.003 | 0.011 | 0.003 | 0.003 | etn:n | | 0.004 | 0.010 0.004 | 0.002 | 0.004 | 0.008 | 200.0- | 1000 | -0.004 | 0.004 | 0.008 | 0.004 | -0.008 | | 600.0 | | |
| | | | * * | * * | | * * * | **** | de la construction de la | * * * * * * | * * | * | * * * | * * | | 3) 3) 3) | | * * * | * * | *** | * * * | *** | * | | |
| | | 7 <i>t</i> -stat | -6.378 | -2.053 -5760 | 0.346 | -5.391 | -6.282 | | -9.408 | -7.872 | -1.987 | -6.728 | -2.164 5 577 | 110.0- | | -0.130 | -10.942 | -2.142 | -5.034 | -6.638 | | -9.444 -2.205 | | |
| | | Sell 6 S.E | 0.000 | 0.000 | 0.000 | 0.000 | 0.000 | | 0.000 | 0.000 | 0.000 | 0.000 | 0.000 | 0.000 | 0000 | 0.000 | 0.000 | 0.000 | 0.000 | 0.000 | 0000 | 0.000 | | |
| | βΥ | 5 Coefficient | -0.003 | -0.001 | 0.000 | -0.003 | -0.003 | | -0.004 | -0.003 | -0.001 | -0.003 | -0.001 | -0.002 | 100.0 | cn0.0 | -0.005 | -0.001 | -0.004 | -0.003 | 100.0 | | | |
| | BTC/ | | *** | * * | *** | * * * * * * | * | -to the t | *) *) *) | * * * * | *** | * * | * * | | 9 9 9 9 | * * | **** | * * | * * | * * * | *** | * | | |
| | | 4 <i>t</i> -stat | 4.797 | 2.419 5 184 | 2.805 | 4.512 | 0.040 -8.973 | 0000 | -8.823 | 659.7 6.959 | 5.648 | 6.711 | 7.573 | 060.6- | | -9.733 6.256 | 9.227 | 6.147 | 10.492 3.648 | -9.229 | | o.700 3.220 | | |
| | | Buy 3 S.E | 0.001 | 0.003 | 0.005 | 0.001 | 0.000 | | 0.000 | 100.0 | 0.002 | 0.001 | 0.002 | 0.000 | 0000 | 0.002 | 0.001 | 0.002 | 0000 | 0.000 | 100.0 | 0.002 | | |
| Table 9. | | 2 Coefficient | <i>20 MA</i> 0.002 | 0.007 | 0.013 | 0.002 | -0.003 | 50 MA | -0.004 | 0.004 | 0.010 | 0.004 | 0.012 | -0.003 | 150 MA | -0.004 | 0.005 | 0.012 | c00.0 | -0.003 | 200 MA | 000.0 | | |
| Momentum effect: BTC/JPY and BTC/ZAR | | $\frac{1}{TTR}$ | Long-term (1,20,0) | (1,20,1) | (2,20,1) | (5,20,0) | (1,02,0) Intercept | Long-term | (1,50,0) | (1,50,1) (2,50,0) | (2,50,1) | (5,50,0) | (5,50,1) | Intercept | Long-term | (1,150,1) | (2,150,0) | (2,150,1) | (5,150,1) | Intercept | Long-term | (1,200,1) $(1,200,1)$ | | |

| | | | * * * | * | * * | 2-7) | nge | | Tech | inica |
|-------|----------|------------------|-------------------|-------------|-------------|---------------------------------|--|-------------|-----------------|-----------------|
| | 51 | t-stat | -10.352 1 505 | -2.148 | -7.400 | –17.366 irst panel (2 | each excha | | trading profita | rules bility |
| | Sell 319 | S.E | 0.000 | 0.000 | 0.001 | 0.000 nels. The fi | urately for and 200 | | | 27 |
| AR | ŧ | Coefficient | -0.005 | -0.001 | -0.004 | -0.006 wo main pai | 20, 50, 150 s | | | |
| BTC/Z | | - | * * | *** | * * * | d into t | ls are re amely, | | | |
| | 10 | to t-stat | 1.977 | 7 969 | 6.306 | -21.735 le is divide | l sell signa ge rules, n | | | |
| | Buy o | s.E | 0.003 | 0.000 | 0.001 | 0.000 ten the tab | or buy anc ving avera | | | |
| | α | Coefficient | 0.007 | 0.004 | 0.007 | –0.008 le (<i>TTR</i>). Th | l'he results f etch for mov | | | |
| | | | * * * * | * * | * | ding ru | ÆUR. 1 erm str | | | |
| | Ľ | t-stat | -11.439 | -9495 | -2.031 | -5.222 hnical trac | s for BTC, ach long-te | | | |
| | Sell | S.E | - 0000 | 0000 | 0.000 | 0.000 or each tec | lays result id under æ | | | |
| ΡΥ | Ľ | Coefficient | -0.005 | 100.0- | -0.001 | -0.002 ion outputs f | el (8–13) disp s are reporte | | | |
| BTC/ | | | * * * * * * | * * * | * * * | regress | nd pan output | | | |
| | ~ | + t-stat | 8.217 | 11 975 | 8.053 | -10.072 the list of 1 | le the seco regression \$5%, *10 | | | |
| | Buy 3 | SE | 0.000 | 0.001 | 0.000 | 0.000 nn shows | AUD, whi UR). The 1 ***1%, ** | lation | | |
| | c | 2 Coefficient | 0.004 | 110.0 | 0.004 | –0.004 The first colur | ults for BTC/. AUD, BTC/E significance: ' | : Uwn compi | | |
| | - | $_{TTR}^{1}$ | (2,200,0) | (5,200,1) | (5,200,1) | Intercept Note(s): T | reports rest rate (BTC/ ₁ Statistical s | Source(s) | T | `able 9. |

JCMS profits and thus moderate the potential of TA as an analytical tool in the cryptocurrency market.

References

- Alexander, S.S. (1961), "Price movements in speculative markets: trends or random walks", Industrial Management Review, Vol. 2 No. 2, pp. 7-26.
- Alexander, S.S. (1964), "Price movements in speculative markets trends or random walks, number 2", Industrial Management Review, Vol. 5 No. 2, pp. 25-46.
- Baba, Y., Engle, R.F., Kraft, D.F. and Kroner, K.F. (1989), Multivariate Simultaneous Generalized ARCH, Working paper, University of California, San Diego, CA.
- Bariviera, A.F. (2017), "The inefficiency of Bitcoin revisited: a dynamic approach", Economics Letters, Vol. 161, pp. 1-4, doi: 10.1016/j.econlet.2017.09.013.
- Bariviera, A.F., Basgall, M.J., Hasperué, W. and Naiouf, M. (2017), "Some stylized facts of the Bitcoin market", Physica A: Statistical Mechanics and its Applications, Vol. 484 No. 1, pp. 82-90, doi: 10. 1016/j.physa.2017.04.159.
- Bena, J., Ferreira, M.A., Matos, P. and Pires, P. (2017), "Are foreign investors locusts? The long-term effects of foreign institutional ownership", Journal of Financial Economics, Vol. 126 No. 1, pp. 122-146, doi: 10.1016/j.jfineco.2017.07.005.
- Bessembinder, H. and Chan, K. (1995), "The profitability of technical trading rules in the Asian stock markets", Pacific-Basin Finance Journal, Vol. 3 Nos 2-3, pp. 257-284, doi: 10.1016/0927-538X(95) 00002-3.
- Bessembinder, H. and Chan, K. (1998), "Market efficiency and the returns to technical analysis", Financial Management, Vol. 27 No. 2, pp. 5-17.
- Bitcoincharts (n.d.), available at: https://www.Bitcoincharts.com/ (accessed 6 July 2020).
- Borgards, O. (2021), "Dynamic time series momentum of cryptocurrencies", North American Journal of Economics and Finance, Vol. 5 No. 7, 101428, doi: 10.1016/j.najef.2021.101428.
- Borgards, O. and Czudaj, R.L. (2021), "Features of overreactions in the cryptocurrency market", Quarterly Review of Economics and Finance, Vol. 80 No. 1, pp. 31-48, doi: 10.1016/j.gref.2021. 01.010.
- Bouri, E., Gupta, R., Tiwari, A.K. and Roubaud, D. (2017), "Does Bitcoin hedge global uncertainty? Evidence from wavelet-based quantile-in-quantile regressions", Finance Research Letters, Vol. 23 No. 1, pp. 87-95, doi: 10.1016/j.frl.2017.02.009.
- Bouri, E., Lucey, B. and Roubaud, D. (2020), "Cryptocurrencies and the downside risk in equity investments", Finance Research Letters, Vol. 33, p. 101211, doi: 10.1016/j.frl.2019.06.009.
- Brock, W., Lakonishok, J. and Lebaron, B. (1992), "Simple technical trading rules and the stochastics properties of stock returns", The Journal of Finance, Vol. 47 No. 5, pp. 1731-1764, doi: 10.2307/ 2328994.
- Brooks, C. (2014), Introductory Econometrics for Finance, 3rd ed., Cambridge University Press, Cambridge.
- Brush, J.S. and Boles, K.E. (1983), "The predictive power in relative strength & CAPM", The Journal of Portfolio Management, Vol. 9 No. 4, pp. 20-23, doi: 10.3905/jpm.1983.20.
- Chan, K., Hameed, A. and Tong, W. (2000), "Profitability of momentum strategies in the international equity markets", Journal of Financial and Quantitative Analysis, Vol. 35 No. 2, pp. 153-172, doi: 10.2307/2676188.
- Chaum, D. (1983), "Blind signatures for untraceable payments", in Chaum, D., Rivest, R.L. and Sherman, A.T. (Eds), Advances in Cryptology, Springer, Boston, MA, pp. 199-203, doi: 10.1007/ 978-1-4757-0602-4_18.

- Cheah, E.T. and Fry, J. (2015), "Speculative bubbles in Bitcoin markets? An empirical investigation into the fundamental value of Bitcoin", *Economics Letters*, Vol. 130 No. 1, pp. 32-36, doi: 10.1016/ j.econlet.2015.02.029.
- Ciaian, P., Rajcaniova, M. and Kancs, D. (2016), "The economics of Bitcoin price formation", *Applied Economics*, Vol. 48 No. 19, pp. 1799-1815, doi: 10.1080/00036846.2015.1109038.
- Cumby, R.E. and Modest, D.M. (1987), "Testing for market timing ability: a framework for forecast evaluation", *Journal of Financial Economics*, Vol. 19 No. 1, pp. 169-189, doi: 10.1016/0304-405X(87)90033-X.
- Dale, C. and Workman, R. (1980), "The arc sine law and the treasury bill futures market", Financial Analysts Journal, Vol. 36 No. 6, pp. 71-74, doi: 10.2469/faj.v36.n6.71.
- Detzel, A., Liu, H., Strauss, J., Zhou, G. and Zhu, Y. (2021), "Learning and predictability via technical analysis: evidence from Bitcoin and stocks with hard-to-value fundamentals", *Financial Management*, Vol. 50 No. 1, pp. 107-137, doi: 10.1111/fima.12310.
- Dyhrberg, A.H. (2016), "Hedging capabilities of bitcoin. Is it the virtual gold?", Finance Research Letters, Vol. 16, pp. 139-144, doi: 10.1016/j.frl.2015.10.025.
- Fama, E.F. and Blume, M. (1966), "Filter rules and stock-market trading", *The Journal of Business*, Vol. 39 No. 1 Part 2, pp. 226-241.
- Fred Database (n.d.), Board of Governors of the Federal Reserve System (US), 3-Month Treasury Bill Secondary Market Rate [DTB3], Retrieved from FRED, Federal Reserve Bank of St. Louis, available at: https://fred.stlouisfed.org/series/DTB3.
- Gandal, N. and Halaburda, H. (2016), "Can we predict the winner in a market with network effects? Competition in cryptocurrency market", *Games*, Vol. 7 No. 3, p. 16, doi: 10.3390/g7030016.
- Garber, P.M. (1990), "Famous first bubbles", Journal of Economic Perspectives, Vol. 4 No. 2, pp. 35-54, doi: 10.1257/jep.4.2.35.
- Gerritsen, D.F., Bouri, E., Ramezanifar, E. and Roubaud, D. (2020), "The profitability of technical trading rules in the bitcoin market", *Finance Research Letters*, Vol. 34 No. 1, 101263, doi: 10. 1016/j.frl.2019.08.011.
- Gerritsen, D.F., Lugtigheid, R.A.C. and Walther, T. (2021), "Can Bitcoin investors profit from predictions by crypto experts?", *Finance Research Letters*, in press, doi: 10.2139/ssrn.3688943.
- Gharaibeh, O.K., Al-Khazali, A. and Al-Quran, A.Z. (2021), "Momentum effect in the Oman stock market over the period of 2005-2018", *The Journal of Asian Finance, Economics and Business*, Vol. 8 No. 2, pp. 711-724, doi: 10.13106/jafeb.2021.vol8.no2.0711.
- Gopane, T.J. (2019), "The interest rate behaviour of bitcoin as a digital asset", in Jallouli, R., Bach Tobji, M., Bélisle, D., Mellouli, S., Abdallah, F. and Osman, I. (Eds), Digital Economy: Emerging Technologies and Business Innovation. Lecture Notes in Business Information Processing, Springer, Cham, Vol. 358, pp. 53-65, doi: 10.1007/978-3-030-30874-2_5.
- Grobys, K., Ahmed, S. and Sapkota, N. (2020), "Technical trading rules in the cryptocurrency market", *Finance Research Letters*, Vol. 32 No. 1, article 101-396, doi: 10.1016/j.frl.2019.101396.
- Harwick, F.C. (2016), "Cryptocurrency and the problem of intermediation", *The Independent Review*, Vol. 20 No. 4, pp. 569-588.
- Hsu, P.H., Hsu, Y.C. and Kuan, C.M. (2010), "Testing the predictive ability of technical analysis using a new stepwise test without data snooping bias", *Journal of Empirical Finance*, Vol. 17 No. 3, pp. 471-484, doi: 10.1016/j.jempfin.2010.01.001.
- IRESS (2017), available at: https://www.iress.com/za/ (accessed 6 July 2020).
- Ito, A. (1999), "Profits on technical trading rules and time-varying expected returns: evidence from Pacific-Basin equity markets", *Pacific-Basin Finance Journal*, Vol. 7 Nos 3-4, pp. 283-330, doi: 10. 1016/S0927-538X(99)00008-6.

Technical trading rules' profitability

| Jacobs, | B.I. | and | Levy, | K.N. | (1988), | "Disentangling | equity | return | regularities: | new | insights | and |
|---------|-------|-------|-------|---------|-----------|-------------------|---------|---------|---------------|---------|------------|--------|
| ir | ivest | ment | oppor | tunitie | es", Find | ancial Analysts J | ournal, | Vol. 44 | No. 3, pp. 18 | -43, do | oi: 10.246 | 9/faj. |
| v | 44.n | 3.18. | | | | | | | | | | |

- Jegadeesh, N. and Titman, S. (1993), "Returns to buying winners and selling losers: implications for stock market efficiency", *The Journal of Finance*, Vol. 48 No. 1, pp. 65-91, doi: 10.1111/j.1540-6261.1993.tb04702.x.
- Jensen, M.C. and Benington, G.A. (1970), "Random walks and technical theories: some additional evidence", *The Journal of Finance*, Vol. 25 No. 2, pp. 469-482, doi: 10.2307/2325495.
- Kho, B.C. (1996), "Time-varying risk premia, volatility, and technical trading rule profits: evidence from foreign currency futures markets", *Journal of Financial Economics*, Vol. 41 No. 2, pp. 249-290, doi: 10.1016/0304-405X(95)00861-8.
- Khuntia, S. and Pattanayak, J.K. (2018), "Adaptive market hypothesis and evolving predictability of bitcoin", *Economics Letters*, Vol. 167 No. 1, pp. 26-28, doi: 10.1016/j.econlet.2018.03.005.
- Kristoufek, L. and Vosvrda, M. (2018), "Herding, minority game, market clearing and efficient markets in a simple spin model framework", *Communications in Nonlinear Science and Numerical Simulation*, Vol. 54 No. 1, pp. 148-155, doi: 10.1016/j.cnsns.2017.05.025.
- Levy, R.A. (1967), "Random walks: reality or myth", *Financial Analysts Journal*, Vol. 23 No. 6, pp. 67-77, doi: 10.2469/faj.v23.n6.69.
- Masry, M. (2018), "The impact of unsystematic risk on stock returns in an Emerging Capital Markets (ECM's) country: an empirical study", *International Journal of Financial Research*, Vol. 9 No. 3, pp. 91-107, 2017:i:3:p:91-107, available at: https://EconPapers.repec.org/RePEc:ibn:ijefaa:v:9:y.
- Miller, N., Yang, Y., Sun, B. and Zhang, G. (2019), "Identification of technical analysis patterns with smoothing splines for bitcoin prices", *Journal of Applied Statistics*, Vol. 46 No. 12, pp. 2289-2297, doi: 10.1080/02664763.2019.1580251.
- Nazário, R.T.F., Silva, J.L.E., Sobreiro, V.A. and Kimura, H. (2017), "A literature review of technical analysis on stock markets", *The Quarterly Review of Economics and Finance*, Vol. 66 No. 1, pp. 115-126, doi: 10.1016/j.qref.2017.01.014.
- Neely, C., Weller, P. and Dittmar, R. (1997), "Is technical analysis in the foreign exchange market profitable? A genetic programming approach", *Journal of Financial and Quantitative Analysis*, Vol. 32 No. 4, pp. 405-426, doi: 10.2307/2331231.
- Neely, C.J., Weller, P.A. and Ulrich, J.M. (2009), "The adaptive markets hypothesis: evidence from the foreign exchange market", *Journal of Financial and Quantitative Analysis*, Vol. 44 No. 2, pp. 467-488, doi: 10.1017/S0022109009090103.
- Neely, C.J., Rapach, D.E., Tu, J. and Zhou, G. (2014), "Forecasting the equity risk premium: the role of technical indicators", *Management Science*, Vol. 60 No. 7, pp. 1772-1791, doi: 10.1287/mnsc. 2013.1838.
- Newey, W.K. and West, K.D. (1987), "Hypothesis testing with efficient method of moments estimation", *International Economic Review*, Vol. 28 No. 3, pp. 777-787, doi: 10.2307/2526578.
- Nugroho, B.A. (2021), "Dynamic risk-based optimization on cryptocurrencies", Journal of Capital Markets Studies, Vol. 5 No. 1, pp. 28-48.
- Park, C.H. and Irwin, S.H. (2007), "What do we know about the profitability of technical analysis?", Journal of Economic Surveys, Vol. 21 No. 4, pp. 786-826, doi: 10.1111/j.1467-6419.2007.00519.x.
- Ratner, M. and Leal, R.P.C. (1999), "Tests of technical trading strategies in the emerging equity markets of Latin America and Asia", *Journal of Banking and Finance*, Vol. 23 No. 12, pp. 1887-1905, doi: 10.1016/S0378-4266(99)00042-4.
- Resta, M., Pagnottoni, P. and De Giuli, M.E. (2020), "Technical analysis on the Bitcoin market: trading opportunities or investors pitfall?", *Risks*, Vol. 8 No. 2, article 44, doi: 10.3390/risks8020044.
- Sensoy, A. (2019), "The inefficiency of Bitcoin revisited: a high-frequency analysis with alternative currencies", *Finance Research Letters*, Vol. 28 No. 1, pp. 68-73, doi: 10.1016/j.frl.2018.04.002.

JCMS 6.1

- Sermpinis, G., Hassanniakalager, A., Stasinakis, C. and Psaradellis, I. (2021), "Technical analysis profitability and persistence: a discrete false discovery approach on MSCI indices", *Journal of International Financial Markets, Institutions and Money*, Vol. 73, 101353, doi: 10.1016/j.intfin. 2021.101353.
- Shakya, V., Pavar Kumar, P. and Tewari, L. (2021), "Blockchain-based cryptocurrency scope in India", Proceedings – 5th International Conference on Intelligent Computing and Control Systems (ICICCS) 2021, Madurai, India, IEEE, 6-8 May, Piscataway, NJ, pp. 361-368, doi: 10.1109/ ICICCS51141.2021.9432143.
- Smidt, S. (1965), "A test of the serial independence price changes in soybean futures", Food Research Institute Studies, Vol. 5 No. 2, pp. 117-136.
- Sweeney, RJ. (1986), "Beating the foreign exchange market", *The Journal of Finance*, Vol. 41 No. 1, pp. 163-182, doi: 10.1111/j.1540-6261.1986.tb04497.x.
- Sweeney, R.J. (1988), "Some new filter rule tests: methods and results", The Journal of Financial and Quantitative Analysis, Vol. 23 No. 3, pp. 285-300, doi: 10.2307/2331068.
- Taylor, S.J. (1986), Modelling Financial Time Series, John Wiley, Chichester.
- Tiwari, A.K., Jana, R.K., Das, D. and Roubaud, D. (2018), "Informational efficiency of Bitcoin –an extension", *Economics Letters*, Vol. 163 No. 1, pp. 106-109, doi: 10.1016/j.econlet.2017.12.006.
- Ülkü, N. and Prodan, E. (2013), "Drivers of technical trend-following rules' profitability in world stock markets", *International Review of Financial Analysis*, Vol. 30 No. 1, pp. 214-229, doi: 10.1016/j. irfa.2013.08.005.
- Urquhart, A. (2016), "The inefficiency of Bitcoin", *Economic Letters*, Vol. 148 No. 1, pp. 80-82, doi: 10. 1016/j.econlet.2016.09.019.
- Urquhart, A., Gebka, B. and Hudson, R. (2015), "How exactly do markets adapt? Evidence from the moving average rule in three developed markets", *Journal of International Financial Markets, Institutions and Money*, Vol. 38 No. 1, pp. 127-147, doi: 10.1016/j.intfin.2015.05.019.
- Van Horne, J.C. and Parker, G.C.C. (1967), "The Random-Walk Theory: an empirical test", *Financial Analysts Journal*, Vol. 23 No. 6, pp. 87-92, doi: 10.2469/faj.v23.n6.87.

Technical

trading rules'

profitability

| JCMS 6,1 | Appendix | |
|---------------|----------------|---|
| | No | Technical trading rules (short, long, band) |
| | 1 | (1, 20, 0) |
| 32 | 2 | (1, 20, 1) (1, 20, 1) |
| | 3 | (2, 20, 0) |
| | 4 | (2, 20, 1) |
| | 5 | (5, 20, 0) |
| | 6 | (5, 20, 1) |
| | 7 | (1, 50, 0) |
| | 8 | (1, 50, 1) |
| | 9 | (2, 50, 0) |
| | 10 | (2, 50, 1) |
| | 11 | (5, 50, 0) |
| | 12 | (5, 50, 1) |
| | 13 | (1, 150, 0) |
| | 14 | (1, 150, 1) |
| | 15 | (2, 150, 0) |
| | 16 | (2, 150, 1) |
| | 17 | (5, 150, 0) |
| | 18 | (5, 150, 1) |
| | 19 | (1, 200, 0) |
| | 20 | (1, 200, 1) |
| | 21 | (2, 200, 0) |
| | 22 | (2, 200, 1) |
| Table A1. | 23 | (5, 200, 0) |
| Technical | 24 | (5, 200, 1) |
| Trading Rules | Source(s): Own | |

Corresponding author Thabo J. Gopane can be contacted at: tjgopane@uj.ac.za

For instructions on how to order reprints of this article, please visit our website: www.emeraldgrouppublishing.com/licensing/reprints.htm Or contact us for further details: permissions@emeraldinsight.com